ECONSTOR Make Your Publications Visible.

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Sonmez, Sinem

Article

The dynamic effects of oil price shocks on the economies of the twelve major oil exporting countries from 1970-2013: The role of political economy factors

International Journal of Business and Economic Sciences Applied Research (IJBESAR)

Provided in Cooperation with:

International Hellenic University (IHU), Kavala

Suggested Citation: Sonmez, Sinem (2020) : The dynamic effects of oil price shocks on the economies of the twelve major oil exporting countries from 1970-2013: The role of political economy factors, International Journal of Business and Economic Sciences Applied Research (IJBESAR), ISSN 2408-0101, International Hellenic University (IHU), Kavala, Vol. 13, Iss. 1, pp. 29-51, https://doi.org/10.25103/ijbesar.131.04

This Version is available at: https://hdl.handle.net/10419/242222

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.



https://creativecommons.org/licenses/by-nc/4.0/

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



WWW.ECONSTOR.EU



International Journal of Business and Economic Sciences Applied Research

The Dynamic Effects of Oil Price Shocks on the Economies of the Twelve Major Oil Exporting Countries from 1970-2013: The Role of Political Economy Factors

Sinem Sonmez*

New School University, Department of Economics, New York, NY

| ARTICLE INFO | ABSTRACT |
|---------------------------|--|
| Article History | Purpose: |
| | This paper examines whether the economies of oil-exporting countries respond to oil shocks |
| Received 28 January 2020; | differently, depending on the country's political economy factors, such as regional economic |
| Accepted 15 April 2020 | alliance, stage of economic development, and the exchange rate regime, using a structural |
| P16 E51 E31 O41 | Vector Error Correction Model (VECM). The sample countries covered in this study |
| 110, 201, 101, 211 | include all the major twelve oil exporters: the GCC countries (Bahrain, Kuwait, Oman, |
| | Saudi Arabia, UAE, and Qatar)1 and the non-GCC oil exporters (Iran, Nigeria, Norway, |
| | Canada, Russia and Venezuela) for the 1970-2013 period. |
| | Design/methodology/approach: |
| | To achieve the above, this paper employs a four-variable restricted structural Vector Error |
| | Correction Model (VECM) with oil prices (exogenous) and a set of endogenous variables, |
| | including GDP, M2, and Inflation for the 1970-2013 period, with a co-integrating |
| | relationship that varies from 1 to 2. Both Johansen Cointegration Test and Granger |
| | Causality Tests have been applied. To eliminate the effects of a contemporaneous correlation |
| | of the residuals with the regressors, the errors are orthogonalized by a Choleski |
| | decomposition. |
| | Finding: |
| | The political economy characteristics of countries explain why the economies of oil |
| | exporters either with fixed or managed floating exchange rates behave differently than |
| | those economies with floating exchange rates. Money is endogenous in oil-exporting |
| | countries with fixed or managed floating exchange rates, but not in oil-exporting countries |
| | with floating exchange rates. The evidence found in this paper suggests that the fixed |
| | exchange rate regime of the GCC countries is not effective in tamping down inflationary |
| Keywords: | pressures, as planned. There is a strong positive effect of oil price shocks on the foreign |
| Macroeconomics, | exchange reserves of the GCC countries, whereas the effect of oil price shocks on foreign |
| Fiscal Policy Oil Prices | exchange reserves has been rather non-existent for the non-GCC oil exporters with floating |
| Oil Exporters | exchange rates. |
| | Research limitations/implications: |
| | The data span is restricted by data availability, the study could ensure for more robustness |
| | and better confidence with quarterly data, but most of the variables for countries are |
| | reported annually, except for Canada, Russia, and Norway. |
| | Originality/value: |
| | This paper separates oil exporters into two distinct categories, the GCC countries and the |
| | non-GCC oil exporters. To my knowledge, this has not been done before in the academic |
| | literature. Unlike the GCC countries, the non-GCC oil exporters do not participate in a |
| | common political alliance or an economic union. Their economies are diversified and less |
| | dependent on oil exports. This study shows that the political economy characteristics play a |
| | large role in explaining why the economies of oil exporters with either fixed or managed |
| | floating exchange rates behave differently than the oil exporters with floating exchange |
| | rates. |
| | @International Hellenic University |
| | ©International Referic University |

¹ The currencies of the GCC countries, other than that of Kuwait, are formally pegged to the US dollar.

[†]Corresponding Author: SINEM SONMEZ Email: sonms376@newschool.edu

DOI: 10.25103/ijbesar.131.04

1. Introduction

The economics of the oil-exporting countries respond differently to oil shocks depending on the political economic factors, such as regional economic alliance, stage of economic development, and the exchange rate regime. The purpose of this paper is to explore whether political economy factors is a reason why the macroeconomic variables of oil exporters behave differently in response to oil shocks. I am particularly interested in understanding the behaviour of economic variables, such as real government spending, real output, real exchange rates, foreign exchange reserves, reserve money, and the money supply, in response to oil price shocks.

Recent papers focused on advanced economies, which are importers of oil rather than on countries that export oil. Unlike the papers that focus primarily on industrialized economies, here the focus is entirely on a group of oil exporters. This paper focuses on the macroeconomic responses of all twelve major oil exporters to oil price shocks, rather than focusing on the macroeconomic response of a select few oil exporters or advanced economies that import oil, and separates the major oil exporters into two distinct categories, the GCC countries and the non-GCC oil exporters. To my knowledge this has not been done before in the academic literature. This categorization of oil exporters is important, since the GCC countries share a political and economic alliance and their economies are largely oil dependent, with currencies (other than that of Kuwait) formally pegged to the US dollar. Unlike the GCC countries, the non-GCC oil exporters do not participate in a common political alliance or an economic union. Their economies are diversified and less dependent on oil exports. The GCC countries and non-GCC oil exporters combined are known to account for more than 50 percent of the global oil reserves and exports, according to the US Energy Information Administration (EIA).

This paper aims to address the following set of policy questions for all the twelve major oil exporters, in response to oil price shocks, explaining in the context of political economy factors, as mentioned above:

Are fiscal policies of oil exporters pro-cyclical or countercyclical? Is money endogenous for oil exporters? Do real oil price shocks have an impact on the real exchange rates of oil-exporting countries, and if so, does a real exchange rate appreciation come in the form of an increase in the price level or an appreciation of the nominal exchange rate? Do foreign exchange reserve shocks (through the transmission of oil prices) have a positive effect on reserve money and subsequently the money supply?

These are questions of key importance and have important policy relevance for the twelve major oilexporting countries.

Oil-exporting countries do not form a homogenous group. There is a significant variation in the extent of a country's oil dependence. In some countries, oil accounts for majority of exports, while in other countries oil plays a less significant role.

The economic intuition suggests that oil is an international commodity, whose price is determined by the global supply and demand conditions. Thus, small economies such as the ones I study in this paper are unlikely to affect the price of oil on their own, possibly except for Saudi Arabia, which is a leading member of OPEC that influences the price of oil.

The identification of oil as an exogenous variable would be incorrect if economic developments in the country under consideration would affect the world price of oil contemporaneously, but the countries examined in this paper are all oil exporters and not the advanced economies that would normally affect the price of oil. (Blanchard and Gali, 2007).

The importance of oil in each economy varies widely across the sample of oil exporters. In relation to an overall GDP, the share of oil exports ranges from about 4 percent in Canada, 11 percent in Norway and Russia, and nearly to 50 percent in the case of Bahrain, Kuwait, and Oman.

I make several other contributions to this paper. I treat real oil shocks as exogenous, as substantiated by Granger causality test results and the interpretation of variance decomposition results from the Vector Error Correction Models, such a treatment not previously reported in the academic literature results in a significantly high R2 for all the structural VECM equations. I would like to add that the literature of this paper, which explores similar macroeconomic questions, treats oil as endogenous. (See, Bernanke, Gertler and Watson (1997), Blanchard and Gali (2007), Kilian (2009), and Kilian (2010)). Second, this paper complements the existing large literature on fiscal shocks being exogenous with respect to output, (see, Monacelli and Perotti (2006), Ravn. et al. (2012), Enders et. al (2011)), with results that are virtually the same when there is a change in the order of variables and that lags of real output do not Granger-cause real government spending. Similarly, lags of M2 do not Granger-cause GDP for any of the oil exporters and the results are insensitive to a change in the order of variables, which suggest that the existing order of variables from GDP to M2 to Inflation remains correct for the twelve major oil exporters of this paper. This is not something which has been done before in academic literature.

Some of the main highlights of this paper are discussed below and the rest of the findings are discussed in Section 2. The findings of this paper are reasonably close to the findings of empirical literature, also discussed in Section 2.

The literature survey suggests that the effects of fiscal policy shocks on output or the effects of output shocks (through the transmission of oil prices) on fiscal policy have not been studied together for the major oil-exporting countries. I aim to contribute to this part of the literature.

First, this paper explores whether the fiscal spending shocks affect output, since the structural VECMs show that there is not a positive estimated response of real economic activity to real government spending shocks in eight out of the ten oil exporters (out of the ten countries with available government spending data), except for Iran and Canada, which means that there is not a countercyclical fiscal policy response in most the twelve-major oil-exporting countries. The non-existence of a response of real economic activity to real fiscal spending shock is not surprising, but rather consistent with findings from academic literature that the fiscal policies of developing countries tend not to be countercyclical, but rather they are pro-cyclical, as is shown to be the case here for Saudi Arabia, Oman, Bahrain, and Venezuela.

Second, the literature survey also suggests that the effects of real oil prices on real exchange rates have been studied relatively little. I aim to contribute to this part of the literature.

Since oil constitutes a large portion of the revenues of oil exporters, especially of the GCC countries examined here, real oil prices affect their real exchange rates. There is no real exchange rate appreciation from real oil shocks in any of the non-GCC oil exporters, except for Venezuela, which may be because oil accounts for a small share of output, or simply that these countries may have more monetary policy options available under a floating/managed floating exchange rate regime than those available under a fixed exchange rate regime.

Later, I investigate the effect of real oil shocks on the real exchange rates of oil exporters, where I determine whether the real exchange rate appreciation takes the form of inflation or a nominal exchange rate appreciation, which sharply differentiates this paper from earlier studies. The theory predicts that a real exchange rate appreciation under fixed exchange rate regimes comes primarily in the form of inflation instead of a nominal exchange rate appreciation. This paper concludes that a real exchange rate appreciation from real oil price shocks in the GCC countries and Venezuela, with fixed/managed floating exchange rates, comes in the form of a rise in the price level, with virtually no change in their nominal exchange rates consistent with economic theory.

The real exchange rate appreciation from real oil price shocks could have broad policy implications for oil exporters, since a real exchange rate appreciation can deteriorate the countries' trade and current account balances, which may adversely affect their non-oil sector and make the non-oil sector become less competitive in global markets.

Third, this paper discusses the presence of endogenous money, which goes from economic activity to money supply. This paper shows that money is endogenous in oil-exporting countries with fixed or managed floating exchange rates, including the GCC countries, Venezuela, and Iran, but money is not endogenous in countries with floating exchange rates, such as Canada, Norway, and Nigeria1, since it is the exchange rate that responds to an external shock, such as GDP, rather than the money supply.

Fourth, this paper shows that foreign exchange reserve shocks (through the transmission of oil prices) have a strong positive effect on the reserve money of the GCC countries and in Venezuela, Nigeria, and Iran, countries with either fixed or managed floating exchange rates. In Canada, Norway, and Nigeria, with a purely floating exchange rate regime, the response of reserve money to foreign exchange shocks has been mostly non-existent, either with weak or no impact of foreign exchange reserves on reserve money, consistent with economic reasoning.

Sixth, this paper suggests that there is a positive effect of the changes in oil prices on the money supply (M2) of Oman, Nigeria2, Venezuela, and Russia, either with fixed or a managed floating exchange rate, but there is no change in the money supply (M2) to changes in the oil prices for oil-exporting countries with floating exchange rates, such as Canada and Norway.

This paper proceeds as follows: Section 2 provides a literature survey. Section 3 presents the basic facts. Section 4 presents the methodology. Section 5 discusses the response of real output from real government shocks, the response of real exchange rates to real oil shocks, the response of reserve money to foreign exchange reserve shocks, and finally, the response of money supply to changes in oil prices. Section 6 concludes.

2. Literature Review

This paper relates to several strands of literature.

This paper contributes to the dearth of academic literature on the effect of fiscal shocks on output in the developing countries, from the perspective of oil exporting countries.

In contrast to the logic and approach used by Bernanke and Mihov (1998) on monetary policy using the structural VAR model, where identification is achieved by assuming that variables like GDP and interest rates do not react to policy variables such as fiscal policy contemporaneously, this paper follows the same logic proposed by Blanchard and Perotti (2002), which uses a structural VECM method with exogenous fiscal shocks and traces their dynamic effects on real GDP, a method better suited to the study of fiscal policy than perhaps for monetary policy.

Both the Keynesian and standard neoclassical models predict that an increase in government spending has a positive effect on output. Several authors have assessed the effects of fiscal shocks on economic activity, including Rotemberg and Woodford (1992), Fatas and Mihov (2001), Eichenbaum et. al (2003), Blanchard and Perotti (2002), Burnside et al. (2004), Perotti (2005), and Mountford and Uhlig (2008), who discuss the countercyclical nature of fiscal policy of advanced economies, where fiscal policy is exogenous with respect to output and has a positive effect on output.

¹ Nigeria adopted a floating exchange rate in 1986.

² Nigeria had a fixed exchange rate regime from 1970-1985, with naira pegged to the dollar, and in 1986, the country adopted a flexible exchange rate.

As in Anshasy (2014), the structural VECM models show that the effects of fiscal policy on the economic activity of oil exporters are from an increase in oil revenues, and for a group of highly oil-dependent countries, fiscal spending is tied to oil revenues, which has a significant positive effect on economic growth, but there is no fiscal spending from oil revenues in less oil-dependent economies.

More generally, this paper builds on the vast literature related to the effects of fiscal spending shocks on output, similar to that of Rotemberg and Woodford (1992), Fatas and Mihov (2001), Eichenbaum et. al (2003), Blanchard and Perotti (2002), Burnside et al. (2004), Perotti (2005), and Mountford and Uhlig (2008), who treat government spending exogenous with respect to output. This paper differentiates itself from other papers, since the four-variable structural VECM results are largely insensitive to the ordering and GDP does not predict M2.

Close to the analysis of this paper on real exchange rates, there are several other papers looking at the effects of real oil price shocks on the real exchange rates, with an investigation into the behavior of real exchange rates under a fixed exchange rate regime and a floating exchange rate.

Methodologically, the empirical strategy of this paper is similar to that of Mussa (1976) and Frankel (2010), with an investigation into the behavior of real exchange rates under two different exchange rate regimes: fixed exchange rate regime, where the nominal exchange rate between the two countries is kept fixed or within narrow bands, and a floating exchange rate regime where market forces are allowed significant latitude to move the nominal exchange rate on a continual basis. Mussa (1976) and Frankel (2010) show that there are substantial and systematic differences in the behavior of real exchange rates under two different exchange rate regimes, and the increased variability of real exchange rates, with little or no contribution from changes in the variability of ratios of national price levels. On the other hand, the variability of real exchange rates under fixed exchange rate regimes has to do with strictly the variability of ratios of price levels.

Many papers discuss the presence of endogenous money, where economic activity affects the money supply. One of the contributions of this paper is to show how output shocks affect the money supply through the Vector Error Correction Models. Marx (1907), Nell (1967), and Minsky (1990) study the effects of output on the money supply, including how money supply depends on profit seeking activity, i.e., money is endogenous, how nominal aggregate demand determines money, and how money correlates with the preferred measure of aggregate demand.

This paper complements the existing literature on the effects of a commodity boom on foreign exchange reserves and base money, with an emphasis given to the role of an exchange rate, fixed vs. floating. Here, I am interested in oil-exporting countries and the paper focuses on the impact of oil price shocks on foreign exchange reserves and reserve money using structural Vector Error Correction Models.

The results of this paper are much like that of Cuddington (1989) and Krugman and Obstfeld (2006), who assert that the central bank's foreign reserves surge at the outset of a commodity boom. Unless the monetary authority uses effective mechanisms for sterilizing the monetary impact, the inflow in foreign exchange reserves will cause the domestic monetary base and the broader monetary aggregates to expand.

In countries with fixed or managed floating exchange rates, an increase in foreign exchange reserves leads to an equal amount of an increase in the money supply, whereas a decrease in foreign exchange reserves leads to a decline in the money supply. In countries with floating exchange rates, foreign exchange reserves do not affect reserve money, consistent with the economic intuition.

More generally, there is vast literature out there on the possible effects of external shocks, such as GDP on the money supply; however, here the focus is entirely on the effect of oil shocks on the money supply, which vary depending on the exchange rate, fixed vs. flexible.

The results of this paper are similar to that of Mundell (1963) and Cuddington (1989), who discuss the economic effects of a fixed exchange regime, where an exogenous increase in GDP or any other external shock forces the monetary authority to supply the market with local currency to keep the exchange rate unchanged. Under a floating exchange rate, however, any exogenous changes in GDP or any oil price changes are offset by changes to the exchange rate. The results of this paper are also similar to that of Mussa (1976), who argues that when the demand for money rises relative to the supply of money, either the domestic credit component of the money supply expands, the exchange rate appreciates, or else, there is a combination of these effects, depending on the exchange rate regime, fixed vs. flexible.

3. Data and Methodology

3.1. Data

This paper studies the world's top twelve oil exporters. The oil exporters studied in this paper account for more than 50 percent of the global oil reserves and exports, according to the US Energy Information Administration (EIA). Unlike the GCC countries that are part of a political and economic alliance, the rest of the six oil exporters do not have any common regional and economic ties.

Figure 1 below illustrates the trend in oil prices from 1970 to 2013 on an annual basis. More specifically, it shows the annual price of a barrel of oil, measured in US dollars. Figure 1 shows a trend for real oil prices for the same period. As Figure 1 shows, changes in real oil prices in log levels have been very large and concentrated over the 1978-1980 and 2000-2008 periods, the latter period known as the most recent run-up in oil prices, with oil prices

that reach their peak in 2008. Oil prices dropped from \$37 to \$19 from 1980 to 1999 then continued their rise from \$30 to \$99 for the 2000-2008 period, following a decline in oil prices to \$61 in 2009, due to the financial crisis, and a sudden increase in oil prices to \$105 in 2012, levelling off at \$97 in 2013. The real US dollar price of oil has been calculated using crude oil prices in US dollars deflated by the US consumer price index, in log levels and first differences.

The countries represented in this paper are from a list of top oil exporters for 2013 by US EIA, which consists of Saudi Arabia, Russia, UAE, Kuwait, Iraq, Nigeria, Venezuela, Qatar, Angola, and Canada, and to this list of oil exporters, Oman, Bahrain, Norway, Iran have been added, while Angola and Iraq have been removed, without the complete economic data.

Based on US EIA's report on top world oil net exporters for 2013, (in thousands of barrels per day), Saudi Arabia exported the largest volume of oil around 11,000 barrels, Russia is the world's second-largest exporter of oil (after Saudi Arabia) with an export of around 10,000 barrels of oil, Canada ranks third with an export of 4,000 barrels of oil, UAE is the world's fourth largest exporter of oil, with an export of oil of about 3,400 barrels of oil, Iran is the world's fifth largest exporter of oil, with 3,000 barrels of oil, Kuwait is the world's sixth largest exporter of oil, with an export of 2,800 barrels of oil, Venezuela is considered the world's seventh-largest oil exporter, with an export of 2,700 barrels of oil, Qatar ranks eighth with an export of 2,000 barrels of oil, Nigeria ranks ninth with an export of oil of around 2,300 barrels of oil, and Norway, Europe's largest oil producer, and the world's tenth exporter of oil, export 1,845 barrels of oil, and lastly, Oman and Bahrain take the eleventh and twelfth spots in a list of major oil exporters, with an export of oil of 945 and 61 barrels of oil per day, respectively.



0.2

1970 1975 1980

2005 2010

Figure 1: Oil Prices in US \$ (1970-2013)

20

٥

1970 1975 1980 1985 1990 1995 2000

Source: IMF (IFS) database, 2014, Crude Oil (Petroleum), US Dollars per Barrel; US Consumer Prices, Index 2005=100.

1985

1990

1995 2000 2005

2010



The exchange rate regimes of the GCC countries and non-GCC oil exporters of this paper are as follows:

The currencies of the GCC countries have been formally pegged to the US dollar. The fixed exchange rates (USD to Local Currency) for Saudi Arabia, Bahrain, and Oman have been fixed at 3.75, 0.38, and 0.38 for one US \$, respectively, from 1986-2013. For Qatar and UAE, their currencies have been fixed at 3.64 and 3.67, respectively, from 1980-2013. Kuwait's currency has been pegged to a weighted currency basket from 1975-2003, and from 2003-2007, their peg has been switched to 0.29 for one US \$, and since 2007, the country's currency has been pegged to a basket of currencies.

Both Canada and Norway have purely floating exchange rates. Since 1999, Russia has implemented a policy of a managed floating exchange rate. The Venezuelan Bolivar is pegged to the U.S. dollar and has been supported by foreign exchange restrictions. Nigeria had a fixed exchange rate regime from 1970-1985, as Nigeria's naira has been pegged to the dollar, and in 1986, the country adopted a flexible exchange rate. Iran's official exchange rate is fixed and officially determined by the government authorities.

Unlike the non-GCC oil exporters, the GCC economies are highly dependent on oil. In the GCC countries, the value of oil exports account on average for 40 percent of the GCC countries' GDP and oil exports range from approximately 36 to 45 percent of GDP for the 1980-20113 period. For the non-GCC oil exporters, oil exports account for approximately 16 percent of GDP for the same period, although their oil exports range from as low as 4 to 11 percent of GDP for Canada, Norway, and Russia, and as high as 39 percent of GDP for Nigeria.

3.2 Methodology

This paper presents a four-variable restricted structural Vector Error Correction Model (VECM) with real oil prices (exogenous) and a set of endogenous variables, including real GDP, real government expenditures, and real exchange rates:

For each of the six countries of the GCC and non-GCC oil exporters, a restricted structural VECM is estimated using four macroeconomic variables: the real US dollar price of oil (in log-differences), real GDP (in log differences), real government expenditures (in log differences), and real exchange rates (in log differences). Each equation in the structural VECM of countries has one lag of the three (endogenous) variables as listed above, contemporaneous real oil prices, affecting each equation, and a constant, using annual data or quarterly data, depending on the availability of data. (See, Enders (2008) for a discussion on VECM method).

In order to eliminate the effects of contemporaneous correlation of the residuals with the regressors, the errors are orthogonalized by Choleski decomposition. (See, Hamilton (1994) and Lutkepohl, H. (2005)). In a Choleski decomposition, all endogenous variables affect each other with lags, with a contemporaneous effect of other endogenous variables, depending on the ordering of variables. Sims (1980), Bernanke (1986), Bernanke and Blinder (1992), Blanchard and Quah (1989) proposed this particular ordering of the innovations in their own structural VAR analysis.

The structural VAR models or VECMs are constructed using quarterly or even monthly data in advanced economies. However, macroeconomic data for the GCC and non-GCC oil exporters exist mostly on an annual frequency, except for Canada, Russia, and Norway, with the available quarterly data. The period covered in this paper is from 1970-2013.

However, this study has some limitations, including the frequency of the data for the oil exporters which is limited to an annual basis for most countries, except for Canada, Russia, and Norway.

The annual data on nominal GDP and Government Expenditures come from the World Development Indicators database (2014) and the data on nominal exchange rates and a country-specific CPI come from IMF's IFS database (2014). For UAE, Oman, and Venezuela, there is no complete CPI data, so GDP deflator (base year varies by country) has been used instead. For Oman, UAE, and Russia, there is no complete data on government spending, which means there is no analysis for these countries.

Real US dollar price of oil is calculated using crude oil prices in US dollars deflated by the US consumer price index. This simple measure reflects yearly changes in crude oil prices. The real government expenditures are calculated from a country's government expenditures deflated by a country's CPI. The Real GDP is calculated from a country's nominal GDP deflated by a country's CPI in log differences. The Real Exchange Rate for each country is calculated using the formula, as discussed below.

The estimated one standard deviation of real oil price innovation is equivalent to 28 percent, using yearly data, and the estimated one standard deviation of real oil price innovation is equivalent approximately to 14 percent, using quarterly data.

The nominal exchange rate, E, reported by IMF IFS data (2014) is (local currency per US dollar), then the definition of a real exchange rate can be expressed as:

$$RER = (P/E)/P^*))$$

(1.2)

That is, E, Nominal Exchange Rate (the price of a unit of foreign currency in terms of local currency) P (the domestic price level)/ P* (the foreign price level), in log differences.

³The data for the value of oil exports are not available prior to 1980 for the GCC countries.

Consumer Price Indices (CPIs) are used to measure national price levels, because reliable data for such price indices are generally available on a yearly and sometimes on a quarterly basis for all the oil exporters covered in this paper for the period studied.

Under this definition, an appreciation of the real exchange rate (an increase in RER) has the same meaning as an increase in the real value of domestic goods in terms of foreign goods.

The structural VECMs, using annual data, is estimated with one lag, given the limited size of the panel data, which is approximately 44 observations for all countries from 1970-2013.

For Russia, Canada, and Norway, countries with quarterly data, Akaike Information Criterion (AIC) and Schwartz Information Criterion (SIC) have been used to determine the number of lags in their structural VECMs. The lag length of the VECM in log levels that minimizes the information criteria are chosen to be three for these three advanced economies.

This paper tests for co-integration of variables using the Johansen co-integration test (1992). Johansen (1988) developed maximum likelihood estimators co-integrating vectors, which provides a rank test to determine the number of co-integrating vectors, r.

For all the oil exporters, there is exogenous oil and a set of endogenous variables, including real government expenditures, real GDP, and real exchange rates, with a co-integrated I (1) process, there is a reduced rank, and a co-integrating relationship that varies from 1 to 2 for each of the vector error correction models of these countries. (See, Engle and Granger (1987)).

Sims (1980), Sims (1986), and Sims, Stock, and Watson (1990) recommend against differencing, since differencing leads to throwing away of information concerning the co-movements in data, when there are co-integrating relationships.

This paper also tests whether lags of real GDP Granger-causes real government spending, based on VEC Granger Causality/Block-Exogeneity Wald Tests and Pairwise Granger Causality Tests. (see, Granger and Newbold (1977)).

Second, this paper presents a four-variable restricted structural Vector Error Correction Model (VECM) with oil prices (exogenous) and a set of endogenous variables, including GDP, M2, and Inflation:

For each of the GCC and non-GCC oil exporters, a restricted structural VECM is estimated using four macroeconomic variables: the nominal US dollar price of oil (in log-differences), nominal GDP (in log differences), M2 (in log differences), and consumer prices (in log differences). Each equation in the structural VECM of countries has one lag of the three (endogenous) variables as listed above, contemporaneous oil prices, affecting each equation, and a constant, using annual or quarterly data, depending on the availability of data. (See, Enders (2008)).

To eliminate the effects of a contemporaneous correlation of the residuals with the regressors, the errors are orthogonalized by a Choleski decomposition. (See, Hamilton (1994) and Lutkepohl, H. (2005)).

The annual data on GDP, M2, and Consumer Price Index are from IMF's IFS database (2014) and World Bank's World Development Indicators database (2014). The quarterly data on GDP, M2, and CPI are from International Monetary Fund's International Financial Statistics (IFS), 2014. For UAE, Oman, and Venezuela, the GDP deflator has been used instead since there is no data available on consumer prices.

The structural VECMs of oil exporters, using annual data, is estimated with one lag given the limited size of panel data, which is of approximately 44 observations from 1970-2013 for these countries.

For Russia, Canada, and Norway, using quarterly data, Akaike Information Criterion (AIC) and Schwartz Information Criterion (SIC) have been used to determine the number of lags in a structural VECM. The lag length of the VECMs in log levels that minimizes the information criteria varies from 1 to 3 for these countries.

For each of the twelve oil exporters, there is exogenous oil prices, and a group of endogenous variables including GDP, M2, and CPI, with a co-integrated I(1) process, where there is a reduced rank, and the co-integrating relationships vary from 1 to 2, and therefore there exists a vector error correction model. (See, Engle and Granger (1987)).

This paper also tests whether lags of M2 Granger-causes GDP, using the VEC Granger Causality/Block-Exogeneity Wald Tests and Pairwise Granger Causality Tests. (See, Granger and Newbold (1977)).

Third, this paper presents a two-variable restricted structural Vector Error Correction Models (VECM) for each of the following set of equations: a). oil prices and real exchange rates. b). oil prices and foreign exchange reserves, c). foreign exchange reserves and reserve money, d). oil prices and inflation.

For each of the twelve oil exporters, the restricted structural VECMs are estimated using oil prices (in logdifferences), real exchange rates (in log differences), foreign exchange reserves (in log differences), reserve money (in log differences), and inflation (in log-differences).

This paper uses data on a quarterly basis and when quarterly data are not available, the annual data are used. The period covered in this paper is from 1970-2013. The quarterly and annual data on oil prices, foreign exchange reserves, and reserve money come from IMF's IFS database (2014).

The estimated one standard deviation of oil price innovation is equivalent to 14 percent, using quarterly data, and the estimated one standard deviation of real oil price innovation is equivalent to 28 percent, using yearly data. The estimated one standard deviation of foreign exchange reserve innovation and one standard deviation of reserve money innovation both vary from country-to-country.

Akaike Information Criterion (AIC) and Schwartz Information Criterion (SIC) have been used to determine the number of lags for structural VECMs, which vary from one to three lags, for the equation on oil prices and inflation. For each of the twelve oil exporters, the oil prices and the endogenous variables including foreign exchange reserves, reserve money, inflation, and real exchange rates, are a co-integrated I(1) process, where there is a reduced rank, and the co-integrating relationships vary from 1 to 2, so there exists a vector error correction model for each equation. (See, Engle and Granger (1987)).

4. The Structural Vector Error Correction Models

4.1 The Impulse Response and Variance Decompositions

Once a restricted structural VECM has been estimated, it can be used as a forecasting tool. In order to investigate the relationships among macroeconomic variables, the impulse response functions (IRFs) are estimated.

The impulse response functions of the restricted structural VECMs illustrate the effect of a temporary shock (a 1-standard deviation) in an endogenous variable, including real government spending, real GDP, and real exchange rates through the dynamic structure of a (VECM) model, while holding all other shocks constant and errors as uncorrelated across variables (Blanchard and Gali, 2007). To understand the importance of a shock in one variable contributing to the variation of another macroeconomic variable, the variance decompositions of each macroeconomic variable are examined.

The variance decompositions are estimated using the Monte Carlo integration methods of 100 replications.

The variance decompositions address the following questions for each oil-exporting country in this paper: How important are real government spending (through the transmission of real oil prices) shocks in accounting for the fluctuations in real output? How important are the real output (through the transmission of real oil prices) shocks in accounting for the fluctuations in real government spending? How important are the real oil price shocks in accounting for the fluctuations in real exchange rates? How important are the output shocks in accounting for the fluctuations of M2? How important are oil shocks in accounting for the fluctuations in foreign exchange reserves, and subsequently, what is the role of foreign exchange reserve shocks in accounting for the fluctuations in reserve money?

The estimated variance decompositions are reported for the sample period of 1970-2013, either over a horizon of ten years or ten quarters, depending on the availability of data.

4.1.1. The Response of Real GDP to Real Government Spending Shocks

This section investigates the effects of a Real Government Spending Shock on Real GDP, where the results vary depending on whether the country is considered a developing or an advanced economy. The fiscal spending shocks are unrelated to unanticipated output shocks, which means that the real government spending shocks are uncorrelated to real output shocks in the structural VECMs. Therefore, the impulse response functions of this paper here capture the direct impact of fiscal spending shocks on real output.

The standard neoclassical, Keynesian, and real business cycle models all predict that an increase in government spending generally has an expansionary effect on economic activity that raises output. The structural VECM results of this paper show that there is no effect of real government spending shocks on real economic activity in eight out of the ten oil exporters, countries with the available data, which means that countercyclical fiscal policy is not employed in any of these countries, except for Iran and Canada.

The non-existence of a response of real economic activity to a fiscal spending shock for majority of the oil exporters is not surprising and consistent with findings from the academic literature that fiscal policies of developing countries tend to be pro-cyclical rather than countercyclical, with ten out of the twelve major oil exporters considered as developing countries.

The structural VECM results here suggest that fiscal spending is countercyclical in both Canada and Iran, since real government spending shocks have a positive effect on real output. The positive response of real output to a real government spending shock in Canada is in line with economic intuition that the fiscal policy response of advanced economies tend to be countercyclical rather than pro-cyclical, in common with Kaminsky, Reinhart, and Vegh (2004) and Lane (2003), who argue that causality runs from fiscal policy to output.

In Iran, there is a strong positive and persistent effect of a real government spending shock on real economic activity. In Canada, there is a relatively small positive effect of a real government spending shock on real economic activity, which are in common with Rotemberg and Woodford (1992), Fatas and Mihov (2001), Blanchard and Perotti (2002), Burnside et al. (2004), and Mountford and Uhlig (2008), who argue that government spending shocks have a positive effect on the economic activity of advanced economies.

In Iran, a real government spending shock leads to a substantial and a persistent rise in real GDP, which accounts for a sizeable 90 percent of the fluctuations in real GDP, whose fiscal policy response is countercyclical. (See, Table 1). The trends in data suggest that there is an inverse relationship between the oil prices and government spending, which highlights the use of a countercyclical fiscal policy employed by the government authorities in Iran.

In Canada, the response of real GDP to a real government spending shock is positive but relatively smaller, with a real government spending shock that accounts for a relatively small 10 percent of the fluctuations in real GDP. (See, Table 1). The positive response of real economic activity to a real government spending shock in Canada is consistent with the conventional wisdom that advanced economies tend to have countercyclical fiscal policies, rather than a procyclical fiscal policy response.

It is important to mention that Canada's economy is similar to that of the US, where there is not a large social safety net and the government has to rely on automatic stabilizers to offset the decline in aggregate demand during times of economic recession.

The fiscal policy response of Saudi Arabia, Oman, Bahrain, Venezuela, whose economies are largely dependent on oil, including Norway, are pro-cyclical, since real government spending responds positively to a real output shock in these countries, where oil prices drive output, and later the econometric causality goes from output to fiscal spending.

| | Saudi Arab | oia | | Kuwait | | | | |
|--------------|------------|--|---|--------------|-----------|-----------|-----------|--|
| 10th Quarter | | | | 10th Quarter | | | | |
| Variation | Real Govt | Real GDP | Real Exch | Variation | Real Govt | Real GDP | Real Exch | |
| Real Govt | 39% | 52% | 9% | Real Govt | 93% | 6% | 1% | |
| Real GDP | 2% | 98% | 0% | Real GDP | 8% | 65% | 27% | |
| Real Exch | 9% | 61% | 30% | Real Exch | 16% | 13% | 71% | |
| No. Obs: | | 42 | | No Obs: | | 40 | | |
| Period: | | 1972-2013 | Kuwait Ioth Quarter Real Govt Real Govt Real GDP Real GDP Real Exch 9% Real GDP 8% 65% 27% 30% Real GDP 8% 65% 27% 30% Real GDP 8% 65% 27% 30% Real Exch 16% 13% 71% No Obs: 40 Period: 1974-2013 Bahrain Real Exch 26% 60% 4% 62% Real Govt 26% 60% 4% 62% Real GDP 5% 90% 5% 42% Real GOV 26% 20% 72% No. Obs: 31 Period: 1982-2012 UAE Real Exch NA NA NA Real GOV NA NA NA NA Real GOV NA NA NA No Obs: 37 Period: 1977-2013 </th | | | | | |
| | Oman | | | | Bahrain | | | |
| 10th Quarter | | | | 10th Quarter | | | | |
| Variation | Real Govt | Real GDP | Real Exch | Variation | Real Govt | Real GDP | Real Exch | |
| Real Govt | 21% | 28% | 51% | Real Govt | 26% | 60% | 4% | |
| Real GDP | 14% | 24% | 62% | Real GDP | 5% | 90% | 5% | |
| Real Exch | 16% | 41% | 42% | Real Exch | 26% | 2% | 72% | |
| No. Obs: | | 41 | | No. Obs: | | 31 | | |
| Period: | | 1972-2013 | | Period: | | 1982-2012 | | |
| | Qatar | 1972-2013 Period: 1982-2012 UAE 10th Quarter | | | | | | |
| 10th Quarter | ~ | | | 10th Quarter | | | | |
| Variation | Real Govt | Real GDP | Real Exch | Variation | Real Govt | Real GDP | Real Exch | |
| Real Govt | NA | NA | NA | Real Govt | NA | NA | NA | |
| Real GDP | NA | 84% | 16% | Real GDP | NA | 96% | 4% | |
| Real Exch | NA | 26% | 74% | Real Exch | NA | 7% | 93% | |
| No. Obs: | | 33 | | No Obs: | | 37 | | |
| Period: | | 1981-2013 | | Period: | | 1977-2013 | | |
| | Canada | | | | Norway | | | |
| 10th Ouarter | | | | 10th Ouarter | · · · · | | | |
| Variation | Real Govt | Real GDP | Real Exch | Variation | Real Govt | Real GDP | Real Exch | |
| Real Govt | 59% | 8% | 33% | Real Govt | 37% | 50% | 13% | |
| Real GDP | 11% | 34% | 55% | Real GDP | 5% | 76% | 19% | |
| Real Exch | 7% | 3% | 90% | Real Exch | 5% | 7% | 88% | |
| No Obs: | | 42 | | No Obs: | | 42 | | |
| Period: | | 1972-2013 | | Period: | | 1972-2013 | | |
| | Venezuel | a | | | Iran | | | |
| 10th Quarter | | | | 10th Quarter | | | | |
| Variation | Real Govt | Real GDP | Real Exch | Variation | Real Govt | Real GDP | Real Exch | |
| Real Govt | 17% | 13% | 70% | Real Govt | 95% | 5% | NA | |
| Real GDP | 4% | 65% | 31% | Real GDP | 92% | 8% | NA | |
| Real Exch | 7% | 20% | 73% | Real Exch | NA | NA | NA | |
| No Obs: | | 42 | | No Obs: | | 35 | | |
| Period: | | 1972-2013 | | Period: | | 1973-2007 | | |
| | Nigeria | | | | | Russia | | |
| 10th Quarter | 2 | | | 10th Quarter | | | | |
| Variation | Real Govt | Real GDP | Real Exch | Variation | Real Govt | Real GDP | Real Exch | |
| Real Govt | 82% | 3% | 15% | Real Govt | NA | NA | NA | |
| Real GDP | 0% | 96% | 4% | Real GDP | NA | 93% | 7% | |
| Real Exch | 58% | 14% | 28% | Real Exch | NA | 0% | 100% | |
| No Obs: | | 31 | | No Obs: | | 76 | | |
| | | | | | | 1995Q1- | | |
| Period: | | 1983-2013 | | Period: | | 2013Q4 | | |

Table 1: Variance Decompositions4 (Real Government Spending, Real GDP, and Real Exchange Rates Shocks and Responses)

4.1.2. The Response of Real Exchange Rates to Real Oil Shocks

This section of the paper first investigates whether there is a real exchange rate appreciation from real oil shocks in all the oil exporters, then later explores whether real exchange rate appreciation in the GCC countries takes

⁴The reported variations are for the 10th period. The shocks are horizontal and the variables that are affected from these shocks are listed vertically.

the form of inflation, since a real exchange rate appreciation under fixed exchange rates is expected to be primarily based on the behavior of domestic price levels, with no contribution of nominal exchange rate to a real exchange rate appreciation, except maybe for sharp changes associated with the official parities, which may affect the real exchange rates, see, Mussa (1976) and Frankel (2010).

The real exchange rate analyses for all the oil exporters are done using quarterly data, except for Oman, Qatar, UAE, and Venezuela, using annual data. For Iran, the real exchange rates are calculated to be zero, so there is no analysis done for the calculation of real exchange rates.

The real oil prices and real exchange rates of the GCC countries and the non-GCC oil exporters are a cointegrated I (1) process, where there is a reduced rank, with one co-integrating relation between real oil prices and the real exchange rates, and therefore there exists a vector error correction model (VECM) for each of the twelve major oil exporters for the 1970-2013 period. (See, Engle and Granger (1987)).

Since oil constitutes a large portion of the GCC countries' revenues, oil prices affect the real exchange rates, particularly in the GCC countries. The real price of oil has been quite volatile in recent decades, so large macroeconomic effects from oil price changes are to occur in the economies of the GCC countries.

There is a real exchange rate appreciation from real oil shocks in the GCC countries, where oil accounts for approximately 40 percent of their economies. (See, Table 2). The real exchange rate appreciation from real oil price shocks in the GCC countries and Venezuela come in the form of a rise in the price level with virtually no change in their exchange rates. The real exchange rate appreciation in these countries from oil price shocks can have broad implications on their trade and current account balances since the non-oil sectors of these countries can become less competitive in the global markets.

The GCC countries instituted a fixed exchange rate regime in 1986, which anchors monetary policy and helps central bank to achieve low-inflation credibility. (See, Frankel (2010)). However, the structural VECM results show that the GCC countries were not able to accomplish their intended objective of lower inflation, which defeats the purpose of having a fixed exchange rate regime.

The positive effect of oil shocks on the inflation of the GCC countries (see, Table 3), as shown in the structural VECMs, lend empirical support to the view that a real exchange rate appreciation of the GCC countries, under fixed exchange rate regimes, comes primarily in the form of inflation and not as a nominal exchange rate appreciation, in common with Mussa (1976) and Frankel (2010). Moreover, there is a full breakdown of the components of a real exchange appreciation for the GCC countries, see Appendix.

The positive effect of the oil shocks on the real exchange rates of the GCC countries are supported by the OLS estimates, which suggest that nominal exchange rates of the GCC countries do not respond to changes in the oil prices, when oil prices are regressed against nominal exchange rates (in log, first differences), and therefore a real exchange rate appreciation in the GCC countries does not take the form of a nominal exchange rate appreciation, other than for sharp devaluations related to the official parities of Saudi Arabia and Oman, which is in common with Mussa (1976) and Frankel (2010). (See, Appendix). The nominal exchange rate graphs for the GCC countries support the OLS estimates, where there have been no significant changes to the nominal exchange rates of the GCC countries, with currencies pegged to the US \$ since 1986, except for Kuwait.

There is no real exchange rate appreciation in response to real oil shocks for any of the non-GCC oil exporters, except for Venezuela. (See, Table 2). For the non-GCC oil exporters, the non-existence in the response of real exchange rates to real oil shocks can be explained by a relatively small share of oil in output, an average of 18 percent from 1970-2011. The nominal exchange rate graphs of the non-GCC oil exporters suggest that the VECM results in countries with managed floating exchange rates, such as Venezuela, Iran, Nigeria, and Russia experience far less volatility in their currencies than that of Norway and Canada, but more volatility than that of the GCC countries, with fixed exchange rates. The volatility of the exchange rates of the GCC countries seems to coincide with either financial or political crises, or else with sudden changes in their nominal exchange rates.

| | Saudi Arabi | a | | Kuwait | / |
|------------------------|-------------------|------------|-----------------------|---|------------|
| | Saudi Arabi | a | 10th Oursel | Nuwait | |
| | Beel Oil | Deel Freek | 10th Quarter | Deal Of | Deel Freed |
| 10th Quarter Variation | | | Variation Bast Oil | | Keal Exch |
| Real Ull Bool Evol | 99% | 1% | Real Ull RealEyel | 94% | 6% 0.6% |
| Kear Exch | 16% | 84% | Na Oha | 4% | 96% |
| NO UDS: | 164 | | NO UDS: | 159 | |
| Danial | 1972Q4- | | Denied | 105401 001000 | |
| Period: | 2013Q3 | | Perioa: | 1974Q1-2013Q3 | |
| | Oman | | | Bahrain | |
| | | D 10 1 | 10th Quarter | | |
| 10th Quarter Variation | <i>i</i> Real Oil | Rea IExch | Variation | Real Oil | Real Exch |
| Real Oil | 93% | 51% | Real Oil | 95% | 5% |
| Real Exch | 87% | 13% | RealExch | 18% | 82% |
| No Obs: | 41 | | No Obs: | 31 | |
| Period: | 1972-2012 | | Period: | 1982-2012 | |
| | Qatar | | | UAE | |
| | | | 10th Quarter | | |
| 10th Quarter Variation | Real Oil | Real Exch | Variation | Real Oil | Real Exch |
| Real Oil | 85% | 15% | Real Oil | 84% | 16% |
| RealExch | 77% | 23% | RealExch | 71% | 29% |
| No Obs: | 33 | | No Obs: | 37 | |
| Period: | 1981-2013 | | Period: | 1977-2013 | |
| | Canada | | | Norway | |
| | | | 10th Quarter | · | |
| 10th Quarter Variation | Real Oil | Real Exch | Variation | Real Oil | Real Exch |
| ~Real Oil | 91% | 9% | Real Oil | 100% | 0% |
| Real Exch | 4% | 96% | RealExch | 6% | 94% |
| No Obs: | 172 | | No Obs: | 174 | |
| Period: | 1971Q1-2013Q4 | | Period: | 1970Q1-2013Q4 | |
| | ~~~~~ | | 1 | ~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~ | |
| | | | | | |
| | Venezuela | | | Russia | |
| 10th Quarter | | | 10th Quarter | | |
| Variation | Real Oil | Real Exch | Variation | Real Oil | Real Exch |
| Real Oil | 60% | 40% | Real Govt | 99% | 1.00% |
| Real Exch | 28% | 72% | Real Exch | 6% | 94% |
| No Obs: | 42 | | No Obs: | 84 | |
| Period: | 1972-2013 | | Period: | 1992Q4-2013Q3 | |
| | Nigeria | | | Iran | |
| 10th Quarter | | | 10th Quarter | | |
| Variation | Real Oil | Real Exch | Variation | Real Oil | Real Exch |
| Real Oil | 93% | 7% | Real Oil | NA | NA |
| Real Exch | 4% | 96% | Real Exch | NA | NA |
| No Obs: | 31 | | No Obs: | | |
| Period: | 1970O1-2013O4 | | Period: | 1970O12013O4 | |

Table 2: Variance Decompositions⁵ (Real Oil and Real Exchange Rates Shocks and their Responses)

 1 The reported variations are for the 10^{th} period. The shocks are horizontal and the variables that are affected from these shocks are listed vertically.

Table 3: Variance Decompositions (Oil and Inflation Shocks and their Impulse Responses)

| | | C . | li Anaki- | | | |
|------------------|-----------|---------------------|------------------|---------|-------------|-----------------|
| | | Sauc | II Aradia | | | |
| 10th Quarter | | | 10th Quarter | | | |
| Variation | Oil | Inflation | Variation | | Oil | Inflation |
| Oil | 98% | 2% | Oil | | 99% | 1% |
| Inflation | 28% | 72% | Inflation | | 27% | 73% |
| No Obs: | | 172 | No Obs: | | | 142 |
| Period | 1971(|)1-901304 | Period: | | 1978(| 0.3-201304 |
| I chiou. | 10112 | <u>201021</u> P. | ahrain | | 1010 | 20 2010 21 |
| | | D | | | | |
| 10th Quarter | | | 10th Quarter | | 0 11 | |
| Variation | Oil | Inflation | Variation | | Oil | Inflation |
| Oil | 98% | 2% | Oil | | 99% | 1% |
| Inflation | 7% | 93% | Inflation | | 37% | 63% |
| No Obs: | | 170 | No Obs: | | | 157 |
| Period: | 19700 |)1-2013Q4 | Period: | | 1979 | Q1-2013Q4 |
| | | K | uwait | | | |
| 10th Quarter | | | 10th Quarter | | | |
| Variation | Oil | Inflation | Variation | | Oil | Inflation |
| | 04.0/ | | v urtation | | | 20/ |
| | 9470 | 0 /0 = 0 0/ | | | 98% | 2 /0 |
| Inflation | 30% | 70% | Initiation | | 20% | 80% |
| No Obs: | | 161 | No Obs: | | | 119 |
| Period: | 1970Q |)1-2013Q4 | Period: | | 1983 | Q2-2009Q3 |
| | Oman | | | | Qatar | |
| 10th Quarter | | | 10th Quarter | | | |
| Variation | Oil | Inflation | Variation | | Oil | Inflation |
| Oil | 94% | 6% | Oil | | 91% | 9% |
| Inflation | 96% | 4% | Inflation | | 45% | 55% |
| No Obs: | 00/0 | 36 | No Obs: | | 1070 | 22 |
| Period: | 10' | 77 0018 | Period: | | 10 | \$1 0018 |
| I citou. | LIAE | 77-2013 | I cilou. | | Durasia 13 | 01-2013 |
| | UAL | | | | Kussia | |
| 10th Quarter | | | 10th Quarter | | | |
| Variation | Oil | Inflation | Variation | | Oil | Inflation |
| Oil | 100% | 0% | Oil | | 76% | 24% |
| Inflation | 85% | 15% | Inflation | | 9% | 91% |
| No Obs: | | 36 | No Obs: | | | 85 |
| Period: | 19' | 77-2013 | Period: | | 1992 | Q2-2013Q4 |
| | | | | | | |
| | | Ca | nada | | | |
| | | | | | | |
| 10th Quarter | | | 10th Quarter | | | |
| Variation | 0:1 | Inflation | Variation | Oil | | Inflation |
| | 70% | a00/ | | 0.00/ | | 1.0% |
| Ull Inflation | 10% | 30% o 7% | Ull Inflation | 8870 | | 12/0 |
| Inflation | 15% | 85% | Inflation | 30% | | 70% |
| | | | | | | _ |
| No Obs: | | 82 | No Obs: | | 87 | 1 |
| Period: | 1970Q1 | I-1991Q1 | Period: | | 1991Q2- | 2013Q4 |
| | | N | orway | | | |
| 10th Quarter | | | 10th Quarter | | | |
| Variation | Oil | Inflation | Variation | Oil | | Inflation |
| Oil | 90% | 0% | Oil | 87% | | 13% |
| Inflation | 9% | 91% | Inflation | 9% | | 91% |
| No Obs: | | 71 | No Obs: | | 91 | l |
| Period: | 197001 | I-1988O2 | Period: | | 198803- | 2013 O 4 |
| | I | ran | | Nigeria | | · ~ · |
| 10th Quantan | 1 | | 10th Quarter | | | |
| Vaniati- | 0:1 | Inflati | Variation | 0:1 | | Inflation |
| v ariation | | initation | v ariation | | | |
| | 99% | 1% | | 99% | | 1% |
| Inflation | 1% | 99% | Inflation | 1% | | 99% |
| No Obs: | 1 | 73 | No Obs: | | 13 | 8 |
| Period: | 1970Q1 | 1-2013Q4 | Period: | | 1970Q1- | 2013Q4 |
| | Venezuela | | | | | |
| 10th Quarter | | | | | | |
| Variation | Oil | Inflation | | | | |
| Oil | 96% | 4% | | | | |
| Inflation | 1% | 99% | | | | |
| No Obsi | 1 /0 | 9970 4.0 | | | | |
| | 10- | T2 | | | | |
| reriod: | 1972 | 2-2013 | | | | |

4.1.3. The Response of Real GDP to Real Government Spending Shocks

This section investigates the effects of a Real Government Spending Shock on Real GDP, where the results vary depending on whether the country is considered a developing or an advanced economy.

The fiscal spending shocks are unrelated to unanticipated output shocks, which means that the real government spending shocks are uncorrelated to real output shocks in the structural VECMs. Therefore, the impulse response functions of this paper here capture the direct impact of fiscal spending shocks on real output.

The standard neoclassical, Keynesian, and real business cycle models all predict that an increase in government spending generally has an expansionary effect on economic activity that raises output. The structural VECM results of this paper show that there is no effect of real government spending shocks on real economic activity in eight out of the ten oil exporters, countries with the available data, which means that countercyclical fiscal policy is not employed in any of these countries, except for Iran and Canada.

The non-existence of a response of real economic activity to a fiscal spending shock for majority of the oil exporters is not surprising and consistent with findings from the academic literature that fiscal policies of developing countries tend to be pro-cyclical rather than countercyclical, with ten out of the twelve major oil exporters considered as developing countries.

The structural VECM results here suggest that fiscal spending is countercyclical in both Canada and Iran, since real government spending shocks have a positive effect on real output. The positive response of real output to a real government spending shock in Canada is in line with economic intuition that the fiscal policy response of advanced economies tend to be countercyclical rather than pro-cyclical, in common with Kaminsky, Reinhart, and Vegh (2004) and Lane (2003), who argue that causality runs from fiscal policy to output.

In Iran, there is a strong positive and persistent effect of a real government spending shock on real economic activity. In Canada, there is a relatively small positive effect of a real government spending shock on real economic activity, which are in common with Rotemberg and Woodford (1992), Fatas and Mihov (2001), Blanchard and Perotti (2002), Burnside et al. (2004), and Mountford and Uhlig (2008), who argue that government spending shocks have a positive effect on the economic activity of advanced economies.

In Iran, a real government spending shock leads to a substantial and a persistent rise in real GDP, which accounts for a sizeable 90 percent of the fluctuations in real GDP, whose fiscal policy response is countercyclical. (See, Table 1). The trends in data suggest that there is an inverse relationship between the oil prices and government spending, which highlights the use of a countercyclical fiscal policy employed by the government authorities in Iran.

In Canada, the response of real GDP to a real government spending shock is positive but relatively smaller, with a real government spending shock that accounts for a relatively small 10 percent of the fluctuations in real GDP. (See, Table 4). The positive response of real economic activity to a real government spending shock in Canada is consistent with the conventional wisdom that advanced economies tend to have countercyclical fiscal policies, rather than a pro-cyclical fiscal policy response.

It is important to mention that Canada's economy is similar to that of the US, where there is not a large social safety net and the government has to rely on automatic stabilizers to offset the decline in aggregate demand during times of economic recession.

The fiscal policy response of Saudi Arabia, Oman, Bahrain, Venezuela, whose economies are largely dependent on oil, including Norway, are pro-cyclical, since real government spending responds positively to a real output shock in these countries, where oil prices drive output, and later the econometric causality goes from output to fiscal spending. (See, Table 4).

| Saudi Arabia | | | | Kuwait | | | |
|------------------------|-----------|---------------|-----------|------------------------|--------------|-----------------------|----------------|
| 10th Quarter Variation | Real Govt | Real GDP | Real Exch | 10th Quarter Variation | Real Govt | Real GDP | Real Exch |
| Real Govt | 39% | 52% | 9% | Real Govt | 93% | 6% | 1% |
| Real GDP | 2% | 98% | 0% | Real GDP | 8% | 65% | 27% |
| Real Exch | 9% | 61% | 30% | Real Exch | 16% | 13% | 71% |
| No Obs: | | 42 | | No Obs: | | 40 | |
| Period: | | 1972-2013 | | Period: | | 1974-2013 | |
| | Oman | | | | Bahrain | | |
| 10th Quarter Variation | Real Govt | Real GDP | Real Exch | 10th Quarter Variation | Real Govt | Real GDP | Real Exch |
| Real Govt | 21% | 28% | 51% | Real Govt | 26% | 60% | 4% |
| Rea lGDP | 14% | 24% | 62% | Real GDP | 5% | 90% | 5% |
| Real Exch | 16% | 41% | 42% | Real Exch | 26% | 2% | 72% |
| No Obs: | | 41 | | No Obs: | | 31 | |
| Period: | | 1972-2013 | | Period: | | 1982-2012 | |
| | Qatar | | | | UAE | | |
| 10th Quarter Variation | Real Govt | Real GDP | Real Exch | 10th Quarter Variation | Real Govt | Real GDP | Real Exch |
| Real Govt | NA | NA | NA | Real Govt | NA | NA | NA |
| Real GDP | NA | 84% | 16% | Real GDP | NA | 96% | 4% |
| Real Exch | NA | 26% | 74% | Real Exch | NA | 7% | 93% |
| No Obs: | | 33 | | No Obs: | | 37 | |
| Period: | | 1981-2013 | | Period: | | 1977-2013 | |
| | Canada | | | | Norway | | |
| 10th Quarter Variation | Real Govt | Real GDP | Real Exch | 10th Quarter Variation | Real Govt | Real GDP | Real Exch |
| Real Govt | 59% | 8% | 33% | Real Govt | 37% | 50% | 13% |
| Real GDP | 11% | 34% | 55% | Real GDP | 5% | 76% | 19% |
| Real Exch | 7% | 3% | 90% | Real Exch | 5% | 7% | 88% |
| No Obs: | | 42 | | No Obs: | | 42 | |
| Period: | | 1972-2013 | | Period: | | 1972-2013 | |
| | Venezuela | D LODD | D 1 D 1 | | Iran | D LODD | D 1 D 1 |
| 10th Quarter Variation | Real Govt | Real GDP | | 10th Quarter Variation | Real Govt | Real GDP | Real Exch |
| Real Govt | 17% | 13% | 70% | Real Govt | 95% | 5% | NA |
| Real GDP | 4% | 65% | 31% | Real GDP | 92% | 8% | NA |
| Na Oba | 170 | 20% | 1370 | No Oba | NA | NA | NA |
| | | 42 | | | | 35 | |
| Period: | 1972-2013 | | | Period: | | 1973-2007 | |
| | Nigeria | B 1055 | D 1 D 1 | | D 1 G | Russia | N 1 N 1 |
| 10th Quarter Variation | Keal Govt | Real GDP | Keal Exch | 10th Quarter Variation | Real Govt | Real GDP | Keal Exch |
| Real Govt | 82% | 3% | 15% | Real Govt | NA | NA | NA |
| Real GDP | 0% | 96% | 4% | Real GDP | NA | 93% | 7% |
| Keal Exch | 58% | 14% | 28% | Keal Exch | NA | 0% | 100% |
| No Obs: | | 31 | | No Obs: | | 76 1995 0 1 | |
| Period: | 1983-2013 | | | Period: | | 2013Q4 | |

Table 4: Variance Decompositions6 (Real Government Spending, Real GDP, and Real Exchange Rates Shocks and their Responses)

⁶The reported variations are for the 10th period. The shocks are horizontal and the variables that are affected from these shocks are listed vertically.

4.1.4. The Responses of M2 to a GDP Shock

This section investigates the effects of a GDP shock on M2, where the results vary depending on the exchange rate regime. The money is endogenous for oil exporters, either with fixed or managed floating exchange rates, as shown in the structural VECMs of oil-exporting countries.

For the GCC countries, the GDP shock leads to an immediate increase in M2. In the GCC countries, M2 gradually increases from a GDP shock, whose effect becomes relatively large in the second year. The GDP shocks explain on average from 59-85 percent of the fluctuations in M2 for over a period of ten year.

| | Saudi Arabia | | | Kuwait | | | |
|---------------------|----------------------|------|-------------|------------------------|-------|---------------------|------|
| 10th Quarter | | | | | | | |
| Variation | GDP | M2 | CPI | 10th Quarter Variation | GDP | M2 | CPI |
| GDP | 81% | 2% | 17% | ~ GDP | 99% | 1% | 0% |
| M2 | 60% | 10% | 30% | M2 | 85% | 14% | 1% |
| CPI | 36% | 1% | 63% | CPI | 10% | 6% | 84% |
| No. Obsi | 3070 | 4.0 | 0370 | No. Obsi | 1070 | 070 | 8470 |
| No. Obs: Poriodi | 1070 0019 | 42 | | No. Obs: Poriod. | | 40 1074 0019 | |
| I eriou; | 1972-2013 Dahasia | | | I erioù: | 0 | 1974-2013 | |
| | Banrain | | | | Omai | n | |
| 10th Quarter | CDD | Ма | CDI | | CDD | Ma | CDI |
| variation | GDP | M2 | | 10th Quarter Variation | GDP | M2 | |
| GDP | 99% | 0% | 0% | GDP | 87% | 12% | 1% |
| M2 | 76% | 20% | 4% | M2 | 85% | 14% | 1% |
| СРІ | 80% | 8% | 12% | CPI | 29% | 8% | 64% |
| No. Obs: | | 37 | | No. Obs: | | 39 | |
| Period: | 1977-2013 | | | Period: | | 1974-2012 | |
| | Qatar | | | | UAE | | |
| 10th Quarter | | | | | | | |
| Variation | GDP | M2 | CPI | 10th Quarter Variation | GDP | M2 | CPI |
| GDP | 92% | 5% | 3% | GDP | 92% | 6.% | 2% |
| M2 | 53% | 33% | 14% | M2 | 75% | 18% | 6% |
| CPI | 11% | 55% | 34% | CPI | 49% | 12% | 39% |
| No Obs: | | 30 | | No. Obs: | | 37 | |
| Period: | 1984-2013 | | | Period: | | 1977-2013 | |
| | Canada | | | | Norwa | ay | |
| 10th Quarter | | | | | | 2 | |
| Variation | GDP | M2 | CPI | 10th Quarter Variation | GDP | M2 | CPI |
| GDP | 97% | 3% | 0% | GDP | 96% | 0% | 4.% |
| M2 | 3% | 96% | 1% | M2 | 23% | 74% | 3% |
| CPI | 26% | 2% | 72% | CPI | 14% | 3% | 83% |
| No. Obs: | 2070 | 170 | ,. | No. Obs: | 11/0 | 149 | 0070 |
| 1101 0 000 | 197103- | 110 | | 110.005. | | 197103- | |
| Period | 901303 | | | Period | | 200803 | |
| I chiou | Venezuela | | | I chibai | Iran | 2000 20 | |
| 10th Quarter | Venezueiu | | | | IIun | | |
| Variation | GDP | Ma | CPI | 10th Quarter Variation | GDP | Ma | CPI |
| GDP | 96% | 1% | 3% | GDP | 78% | 99% | 0% |
| Me | 50% | 4.0% | 1% | Me | 41% | 46% | 1.8% |
| CPI | 07% | 4070 | 1 /0 @0/ | | 15% | 70% | 7.0% |
| No. Obst | 31/0 | 40 | 370 | No. Obsi | 1570 | 170 | 1870 |
| No. Obs: Poriodi | 1070 0019 | 40 | | No. Obs: Poriod. | | 42 1070 0019 | |
| I eriou; | 1970-2013 Nimenia | | | I erioù: | | 1972-2013 Dunain | |
| | Nigeria | | | | | Kussia | |
| 10th Quarter | CDD | Ма | CDI | | CDD | Ma | CDI |
| Variation | GDP | M2 | CPI | 10th Quarter Variation | GDP | M2 | CPI |
| GDP | 38% | 58% | 4% | GDP | 16% | 2% | 82% |
| M2 | 5% | 93% | 2% | M2 | 5% | 36% | 60% |
| | 23% | 37% | 40% | | 0% | 1% | 98% |
| No. Obs: | | 42 | | No. Obs: | | 138 | |
| Period: | 1972-2013 | | | Period: | | 1996-2013 | |

Table 5: Variance Decompositions7 (GDP, M2, and CPI Shocks and their Impulse Responses)

For Iran and Venezuela, M2 also shifts up immediately from a GDP shock. For Iran and Venezuela, the GDP shocks explain a large amount of the fluctuations in M2. Under a fixed or managed floating exchange rate, M2 responds to a

⁷The reported variations are for the 10th period. The shocks are horizontal and the variables affected from those shocks are listed vertically.

GDP shock, whereas there is either a no response or a muted response of M2 to a GDP shock under a floating exchange rate for Canada and Nigeria8, consistent with Mundell (1963) and Mussa (1976), who argue that it is the exchange rate that reacts to an external shock, such as a GDP shock, under a floating exchange rate. In countries with fixed/managed floating exchange rates, the monetary authorities are expected to expand the money supply, which in turn can lead to an increase in output.

The positive responses of M2 to a GDP shock for the GCC countries, Iran, and Venezuela are consistent with Marx (1907), Nell (1967), Nell (2004), who argue that the transaction demand for money from an increase in GDP is met through an increased circulation of money, and similar to that of Minsky (1990), who argues that money supply depends on a profit seeking activity, i.e., money is endogenous.

4.1.5. The Response of Foreign Exchange Reserves to Oil Price Shocks

This section focuses on the effect of oil shocks on foreign exchange reserves.

The restricted structural VECMs have been used for all the twelve oil exporters, with one co-integrating relation between the oil prices and foreign exchange reserves for the 1970-2013 period, and additional VECMs for Qatar and Venezuela have been used, given the structural breaks in the foreign exchange reserves data.

A one standard deviation shock in the oil price (approximately of 14 percent) leads to an immediate shift in foreign exchange reserves for the GCC countries, except for Oman and UAE. For the GCC countries, foreign exchange reserves exhibit a substantial increase from oil price shocks, including for Bahrain, Kuwait, Qatar, and Saudi Arabia, based on the restrictions of a fixed exchange rate regime and given the conditions of a Mundell-Fleming model, which requires central banks to trade domestic currency for foreign currency at a predetermined price, to keep the exchange rate at its preannounced rate.

An oil shock accounts for a large amount of the fluctuations in the foreign exchange reserves of the GCC countries, except for Oman and UAE. The oil shocks account for half of the fluctuations in the foreign exchange reserves of Bahrain and Kuwait, which explain 75 percent of the fluctuations in foreign exchange reserves of Saudi Arabia and Qatar. (See, Table 6).

In Oman and UAE, there is a muted response of foreign exchange reserves to oil shocks, which can be explained by their foreign exchange policies, based on how some of the foreign exchange reserves have been used to purchase foreign securities, where local banks deposit some of their foreign exchange reserves with foreign banks or institutions, or foreign currency reserves are invested in other assets, as shown in their central bank statements.

In contrast to that of the GCC countries, the response of foreign exchange reserves to oil shocks is muted for many of the non-GCC oil exporters, since they have greater leeway for investing their foreign exchange reserves, in the absence of restrictions typically found under a fixed exchange rate regime, except for Nigeria (1970-1983). (See, Table 6).

For the non-GCC oil exporters, including Norway, Nigeria (1984-2009), Russia, Iran, and Venezuela, there is a muted response of foreign exchange reserves to oil shocks. In the absence of any restrictions under a fixed exchange rate regime, these countries are able to use their foreign exchange reserves to purchase foreign securities, deposit some of their foreign exchange reserves with foreign banks or institutions, or else invest their foreign exchange reserves in other assets, including in a money market portfolio, as is the case here for Norway.

In Nigeria (1970-1983), the foreign exchange reserves shift up immediately and stay positive for 10 quarters. An unanticipated shock in oil reserves for Nigeria accounts for more than half of the fluctuations in foreign exchange reserves. In contrast to the 1970-1983 period, the impact of oil prices on foreign exchange reserves is virtually non-existent for the 1984-2009 period. The IRF-based findings on the muted response of foreign exchange reserves to oil shocks are consistent with Kuijs (1998), who argues that the oil production in Nigeria has increased from the 1970s until the early 1980s, thereafter oil prices and oil production both declined.

⁸Nigeria adopted a floating exchange rate in 1986.

| Table 6: | Variance Decom | position Table 9 | Oil, Foreign | Exchange Res | erves Shocks, and | d their Impulse Re | sponses) |
|----------|----------------|------------------|--------------|--------------|-------------------|--------------------|----------|
| | | | 1 - 7 0 | | ,, | | / |

| | ^ | Q | atar | ^ | * / |
|---------------------|-------------|------------------------|------------------------|--------------|----------------|
| 10th Ouarter | | Foreign Exch. | | | Foreign Exch. |
| Variation | Oil | Reserves | 10th Quarter Variation | Oil | Reserves |
| Oil | 95% | 5% | Oil | 99% | 1% |
| Eoroign Eych | <i>JJN</i> | 570 | Foreign Eych | <i>JJ</i> /0 | 1 /0 |
| Foreign Excit. | 6.00% | 04% | Poleigh Excit. | 76.0/ | 24.0/ |
| Keserves | 6.00% | 94% | Reserves | 76% | 24 % |
| No. Obs: | 10 | 97 | No. Obs: | 100 | 44 |
| Period: | 197 | 4Q1-1998Q3 | Break: 1998Q4 | 199 | 8Q4-2009Q3 |
| | Bahrain | | | Kuwait | |
| 10th Quarter | | Foreign Exch. | | | Foreign Exch. |
| Variation | Oil | Reserves | 10th Quarter Variation | Oil | Reserves |
| Oil | 98% | 2% | Oil | 99% | 1% |
| Foreign Exch. | | | Foreign Exch. | | |
| Reserves | 34% | 66% | Reserves | 42% | 58% |
| No Obs: | 01/0 | 173 | No Obs: | | 156 |
| Poriod: | 107 | 17.5 1001_{-2013} | Poriod: | 107 | 1001_200903 |
| Terrou. | 197 C. A | 0Q1-2013Q4 | Teriou. | 197 | 0Q1-2009Q3 |
| 10/1 0 / | 5. Aradia | | | Oman | . |
| 10th Quarter | 0.11 | Foreign Exch. | | 0.11 | Foreign Exch. |
| Variation | Oil | Reserves | 10th Quarter Variation | Oil | Reserves |
| Oil | 100% | 0% | Oil | 99% | 1% |
| Foreign Exch. | | | Foreign Exch. | | |
| Reserves | 78% | 22% | Reserves | 5% | 95% |
| No Obs: | | 148 | No. Obs: | | 136 |
| Period: | 197 | 101-201304 | Period: | 197 | 201-200902 |
| | | UAE | | Norway | ~ ~ |
| 10th Quarter | | Foreign Exch | | Hornuy | Foreign Fych |
| Variation | Oil | Recorder | 10th Quarter Variation | Oil | Recordes |
| o:1 | 08% | 20/ | | 00% | 1 % |
| UII Foreign Fuch | 90 /0 | ∠ /0 | OII Faraian Fuch | 99 /0 | 1 /0 |
| Foreign Excn. | 2.0/ | 070/ | Foreign Excn. | 10/ | 000/ |
| Keserves | 3% | 97% | Reserves | 1% | 99% |
| No. Obs: | | 152 | No. Obs: | | 114 |
| Period: | 197 | 4Q1-2009Q3 | Period: | 197 | 0Q1-2001Q3 |
| | | Russia | | Canada | |
| 10th Quarter | | Foreign Exch. | 10th Quarter | | Foreign Exch. |
| Variation | Oil | Reserves | Variation | Oil | Reserves |
| Oil | 57% | 43% | Oil | N/A | N/A |
| Foreign Exch. | | | Foreign Exch. | 1 | |
| Reserves | 6% | 94% | Reserves | N/A | N/A |
| No Obs: | 070 | 50 | No Obs: | 14/11 | 14/11 |
| Ro Obs. | 1002 | 01,200804 | No Obs. | 10700 | 21 200604 |
| Terrou. | 1993 | Q1-2000Q4 | | 19700 | 21-2000Q4 |
| 1011 0 | | N1 | | | Familian Furt |
| Joth Quarter | 01 | Foreign Exch. | 10th Quarter | 01 | roreign Exch. |
| variation | UII | Keserves | Variation | UII | Keserves |
| Oil | 99% | 1% | Oil | 96% | 4% |
| Foreign Exch. | | | Foreign Exch. | | |
| Reserves | 53% | 47% | Reserves | 4% | 96% |
| No Obs: | | 52 | No Obs: | | 133 |
| Period: | 1970 | Q1-1983Q3 | Break: 1983Q4 | 19830 | Q4-2009Q2 |
| | Iran | | V | /enezuela | |
| 10th Ouarter | | Foreign Exch. | 10th Ouarter | | Foreign Exch. |
| Variation | Oil | Reserves | Variation | Oil | Reserves |
| Oil | 97% | 3% | Oil | 99% | 1% |
| Foreign Evch | 21 10 | 0.10 | Foreign Eych | 2210 | 1/0 |
| Receiver | 2% | 08% | Recerves | 7% | 03% |
| No Oher | ∠ /0 | 90 /0 1 /1 | Ne Oher | 1 /0 | 95 /0 05 |
| NO UDS: | 4050 | 141 | | 40504 | 50 1 001001 |
| Period: | 1970 | Q1-2011Q1 | Period: | 19700 | 21-2013Q4 |

 $^{^{9}}$ The reported variations are for the 10th period. The shocks are horizontal and the variables that are affected from those shocks are listed vertically.

4.1.6. The Response of Reserve Money to Foreign Exchange Reserve Shocks

This section analyses the effect of foreign exchange reserve shocks on reserve money, where results vary depending on the exchange rate regime. At first, this paper identifies the structural breakpoints in the foreign exchange reserves data, using the OLS method.

In the GCC countries, the reserve money exhibits a strong increase from foreign exchange reserve shocks, most notably for Saudi Arabia and Kuwait, consistent with the economic intuition. (See, Table 7).

The trends in data for foreign exchange reserves and reserve money show that reserve money has been consistently below foreign exchange reserves for the 1970-2013 period, which means that only a part of the foreign exchange reserves of the GCC countries has eventually made its way into the reserve money. A part of the foreign exchange reserves in the GCC countries has not been used for internal macroeconomic purposes, i.e., currency or bank deposits, since the government authorities used some of its foreign exchange reserves to either purchase gold, retained some of the foreign exchange reserves as government deposits at the central bank, purchased foreign securities, or local banks deposited some of the foreign exchange reserves with foreign banks or institutions.

The foreign exchange shocks account for about a quarter of the fluctuations in reserve money for Bahrain, Oman, Qatar, and UAE, for the entire 1970-2013 period, and about half of the fluctuations in reserve money for Kuwait and Saudi Arabia from 1999-2013. (See, Table 7).

In support of the VECM results from above, Saudi Arabia's central bank balance statement for 2013 shows that an increase in foreign exchange reserves partially transmits to an increase in broad money, M3, and the rest of the foreign exchange reserves stay as government deposits at the central bank.

In support of the VECM results from above, Kuwait's central bank balance statement for 2013 shows that an increase in foreign exchange reserves partially transmits to an increase in currency and bank deposits, with the rest of the increase in foreign exchange reserves that stays as government deposits. The central bank of Kuwait issues bonds for the remainder of the foreign exchange reserves, which effectively enables the central bank to withdraw liquidity from the financial markets.

There is a positive response of reserve money to foreign exchange reserve shocks for Iran, Venezuela, Russia, and Nigeria, with a managed floating exchange rate, like that of the GCC countries, with fixed exchange rates. (See, Table 6).

The reserve money of Iran, Venezuela, Russia, and Nigeria (1970-1982) shifts immediately from a shock in foreign exchange reserves, where the reserve money remains positive for a protracted period, with half of the foreign exchange reserves of Iran and more than three quarters of the foreign exchange inflows of Venezuela, Russia, and Nigeria (1970-1982) transfer into reserve money. (See, Table 7).

It is important to note that Nigeria had a managed floating exchange rate from 1970-1985, thereafter its exchange rate has been freely floating. With the adoption of a floating exchange rate in 1986, Nigeria's naira has been allowed to devaluate strongly. In 1995, the devaluation was sufficiently large, which subsequently led to a liberalization of the foreign exchange market. For Canada, there has been a loss of foreign exchange reserves from 1979-1999, which explains why there has not been a structural VECM, given the negative values.

The central bank statement for Iran for 2013 support the VECM results from above, which shows that an increase in foreign exchange reserves partially transmits to an increase in currency and bank deposits, where part of the increase in foreign exchange reserves stays as government deposits at the central bank. Iran issues government bonds for the remainder of the currency reserves, which effectively enables the central bank to withdraw liquidity from the financial markets.

The central bank statement for Russia for 2013 support the VECM results from above, which shows that an increase in foreign exchange reserves partially transmits to an increase in currency and bank deposits, and the rest of the increase in foreign exchange reserves transfer into government deposits.

In Norway and Nigeria (from 1983-2009), with floating exchange rates, reserve money hardly changes in response to shocks from foreign exchange reserves, consistent with the economic intuition. In Norway and Nigeria (from 1983-2009), the foreign exchange reserves do not transfer into reserve money, in accord with Mussa (1976) and Mundell (1963), who argue that it is the exchange rate which fluctuates from an external shock, such as oil, and not the base money or the money supply.

Norway invests some of its foreign exchange reserves in an investment fund, which explains why there has not been a positive response of reserve money to foreign exchange reserve shocks. Norway used some of its foreign exchange reserves for internal macroeconomic purposes, i.e., currency or bank deposits, and the rest of its currency reserves has been invested in a global government pension fund, which is a sovereign wealth fund, instead of allowing the foreign exchange reserves to be transmitted to local currency or bank deposits.

In Nigeria, the estimated negative response of foreign exchange reserves to reserve money for the 1983-2009 period may be related to a devaluation at first, and later to a depreciation of the currency, which is associated with a loss of foreign exchange reserves.

| | | Saud | i Arabia | | | | |
|-----------|---------|------------------------|----------------------|---------|-------------------------|--|--|
| 10th | | | | | | | |
| Quarter | Reserve | | 10th Quarter | Reserve | Foreign Exch. | | |
| Variation | Money | Foreign Exch. Reserves | Variation | Money | Reserves | | |
| Reserve | · · | C | | · | | | |
| Money | 74% | 26% | Reserve Money | 43% | 56% | | |
| Foreign | | | | | | | |
| Exch. | | | Foreign Exch. | | | | |
| Reserves | 1% | 99% | Reserves | 1% | 99% | | |
| No Obs: | 1,0 | 172 | No Obs: | 1,0 | 58 | | |
| Period | | | 110 0.001 | | | | |
| (full): | | 197101-201303 | Break: 1999O3 | 199 | 903-201304 | | |
| (1411) | | Bahrain | Kuwait | | | | |
| 10th | | 2000 Will | | | | | |
| Quarter | Reserve | | 10th Quarter | Reserve | Foreign Exch | | |
| Variation | Money | Foreign Exch. Reserves | Variation | Money | Reserves | | |
| Reserve | money | z steign Exem Reserves | , un ution | money | | | |
| Money | 89% | 93% | Beserve Money | 1% | 57% | | |
| Money | 00/0 | 2070 | Reserve Money | 170 | 51/0 | | |
| Foreign | | | | | | | |
| Exch | | | Foreign Exch | | | | |
| Reserves | 1% | 99% | Reserves | 43% | 99% | | |
| No Obs: | 170 | 174 | No Obs: | 10/0 | 157 | | |
| Period | | 197001-901304 | Period | 197 | 001-900903 | | |
| I chiou. | Oatar | 101001 201001 | I chiou | Oman | | | |
| 10th | Zutur | | | 0 maii | | | |
| Quarter | Reserve | | 10th Quarter | Reserve | Foreign Exch | | |
| Variation | Money | Foreign Exch Reserves | Variation | Money | Reserves | | |
| Reserve | money | i oreign Exem neserves | v un ution | money | neserves | | |
| Money | 84% | 16% | Beserve Money | 76% | 94% | | |
| money | 01/0 | 1070 | Reserve money | | 21/0 | | |
| Foreign | | | | | | | |
| Exch. | | | Foreign Exch. | | | | |
| Reserves | 3% | 97% | Reserves | 12% | 92% | | |
| No Obs: | | 141 | No Obs: | | 127 | | |
| Period: | | 1971Q1-2013Q4 | Period: | 197 | 2Q1-2009Q2 | | |
| | | UAE | | | Canada | | |
| 10th | | | | | | | |
| Quarter | Reserve | | 10th Quarter | Reserve | Foreign Exch. | | |
| Variation | Money | Foreign Exch. Reserves | Variation | Money | Reserves | | |
| Poserve | | | | NI / A | NI / A | | |
| Monov | 0.00% | 100/ | Deserve Monor | 11/11 | \mathbf{N}/\mathbf{A} | | |
| money | 90% | 1070 | Reserve Money | | | | |
| Foreign | | | | | | | |
| Exch. | | | Foreign Exch. | | | | |
| Reserves | 3% | 97% | Reserves | N/A | N/A | | |
| No Obs: | | 152 | No Obs: | | | | |
| Period: | | 1974Q1-2009Q3 | Period: | 197 | 003-200804 | | |

Table 7: Variance Decomposition Table10 (Foreign Exchange Reserves, Reserve Money Shocks, and their Responses).

 $^{^{10}}$ The reported variations are for the 10th period. The shocks are horizontal and the variables, which are affected from these shocks, are listed vertically.

| | | | Venezuela | | |
|---|--|---|--|------------------|----------------------------------|
| 10th | | | v chiczucia | | |
| 10ll Quarter | Reserve | Foreign Eych | | Reserve | Foreign Eych |
| Variation | Money | Rosorvos | 10th Quarter Variation | Money | Reserves |
| Reserve | Money | Reserves | Toth Quarter V artation | wioney | Reserves |
| Monov | 570/ | 100/ | Rosomyo Monoy | 70% | 20 ⁰ /- |
| Foreign | 5770 | 4370 | Reserve Money | 7070 | 3070 |
| Fych | | | Foreign Eych | | |
| Posowios | 60/ | 0.4% | Posowos | 4.0/ | 0.6% |
| No Obs. | 070 | 9470 67 | No Obs: | ° F /0 | 9070 |
| Poriod. | 107 | 0.01 1087 0.4 | NO ODS: | 105 | 95 |
| I eriou. | 137 | 021-138724 | Nigoria | 130 | 38Q1-2013Q4 |
| 1.0+h | | | Nigeria | | |
| 10th Overstan | December | Fancian Frah | | Decomio | Fancian Frak |
| Quarter | Manay | Pagamuag | 10th Quantum Vaniation | Monov | Pogowyog |
| V ariation | Money | neserves | Toin Quarter Variation | Money | Reserves |
| Manay | 1 5 0/ | 0.50/ | Becowyo Monoy | 0.00/ | 1.4.0/ |
| Equation | 15% | 83% | Reserve Money | 86% | 14% |
| r oreign Eb | | | Fansian Fashanas | | |
| Excn. | -0/ | 0.20/ | Foreign Exchange | 1 = 0/ | 0.00/ |
| Reserves | 1% | 93% | N _z Ob z | 17% | 83% |
| No Ubs: | 105 | 47 • • • • • • • • • • • • • • • • • • • | | 100 | 103 |
| Period: | 1970 | 001-198201 | Break: 1982Q2 | 198 | 32Q2-2009Q2 |
| 1 | Norwa | у | | Iran | |
| 10th | n | D • D • | | n | D I D I |
| Quarter | Reserve | Foreign Exch. | | Reserve | Foreign Exch. |
| Variation | Money | Keserves | 10th Quarter Variation | Money | Keserves |
| Reserve | | | | | |
| Money | | | | | |
| . . | 87% | 13% | Reserve Money | 54% | 46% |
| Foreign | 87% | 13% | Reserve Money | 54% | 46% |
| Foreign Exch. | 87% | 13% | Reserve Money Foreign Exch. | 54% | 46% |
| Foreign Exch. Reserves | 87% 1% | 13% 99% | Reserve Money Foreign Exch. Reserves | 54%2% | 46% 98% |
| Foreign Exch. Reserves No Obs: | 87% 1% | 13% 99% 112 | Reserve Money Foreign Exch. Reserves No Obs: | 54% 2% | 46% 98% 138 |
| Foreign Exch. Reserves No Obs: Period: | 87% 1% | 13% 99% 112 0Q1-2006Q4 | Reserve Money Foreign Exch. Reserves No Obs: Period: | 54% 2% 197 | 46% 98% 138 20Q1-2010Q4 |
| Foreign Exch. Reserves No Obs: Period: | 87% 1% | 13% 99% 112 0Q1-2006Q4 | Reserve Money Foreign Exch. Reserves No Obs: Period: | 54% 2% 197 | 46% 98% 138 70Q1-2010Q4 |
| Foreign Exch. Reserves No Obs: Period: | 87% 1% 1976 Russia | 13% 99% 112 0Q1-2006Q4 | Reserve Money Foreign Exch. Reserves No Obs: Period: | 54% 2% 197 | 46% 98% 138 70Q1-2010Q4 |
| Foreign Exch. Reserves No Obs: Period: | 87% 1% <u>1970</u> Russia | 13% 99% 112 0Q1-2006Q4 | Reserve Money Foreign Exch. Reserves No Obs: Period: | 54% 2% 197 | 46% 98% 138 70Q1-2010Q4 |
| Foreign Exch. Reserves No Obs: Period: 10th Quarter | 87% 1% <u>1970</u> Russia Reserve | 13% 99% 112 0Q1-2006Q4 Foreign Exch. | Reserve Money Foreign Exch. Reserves No Obs: Period: | 54% 2% 197 | 46% 98% 138 70Q1-2010Q4 |
| Foreign Exch. Reserves No Obs: Period: 10th Quarter Variation | 87% 1% <u>1970</u> Russia Reserve Money | 13% 99% 112 0Q1-2006Q4 Foreign Exch. Reserves | Reserve Money Foreign Exch. Reserves No Obs: Period: | 54% 2% 197 | 46% 98% 138 70Q1-2010Q4 |
| Foreign Exch. Reserves No Obs: Period: 10th Quarter Variation Reserve | 87% 1% 1970 Russia Reserve Money | 13% 99% 112 0Q1-2006Q4 Foreign Exch. Reserves | Reserve Money Foreign Exch. Reserves No Obs: Period: | 54% 2% 197 | 46% 98% 138 70Q1-2010Q4 |
| Foreign Exch. Reserves No Obs: Period: 10th Quarter Variation Reserve Money | 87% 1% <u>1970</u> Russia Reserve Money 99% | 13% 99% 112 0Q1-2006Q4 Foreign Exch. Reserves 1% | Reserve Money Foreign Exch. Reserves No Obs: Period: | 54% 2% 197 | 46% 98% 138 70Q1-2010Q4 |
| Foreign Exch. Reserves No Obs: Period: 10th Quarter Variation Reserve Money Foreign | 87% 1% <u>1970</u> Russia Reserve Money 99% | 13% 99% 112 0Q1-2006Q4 Foreign Exch. Reserves 1% | Reserve Money Foreign Exch. Reserves No Obs: Period: | 54% 2% 197 | 46% 98% 138 70Q1-2010Q4 |
| Foreign Exch. Reserves No Obs: Period: 10th Quarter Variation Reserve Money Foreign Exch. | 87% 1% <u>1970</u> Russia Reserve Money 99% | 13% 99% 112 0Q1-2006Q4 Foreign Exch. Reserves 1% | Reserve Money Foreign Exch. Reserves No Obs: Period: | 54% 2% 197 | 46% 98% 138 70Q1-2010Q4 |
| Foreign Exch. Reserves No Obs: Period: 10th Quarter Variation Reserve Money Foreign Exch. Reserves | 87% 1% <u>1970</u> Russia Reserve Money 99% 90% | 13% 99% 112 0Q1-2006Q4 Foreign Exch. Reserves 1% 10% | Reserve Money Foreign Exch. Reserves No Obs: Period: | 54% 2% 197 | 46% 98% 138 70Q1-2010Q4 |
| Foreign Exch. Reserves No Obs: Period: 10th Quarter Variation Reserve Money Foreign Exch. Reserves No Obs: | 87% 1% <u>1970</u> Russia Reserve Money 99% 90% | 13% 99% 112 0Q1-2006Q4 Foreign Exch. Reserves 1% 10% 50 | Reserve Money Foreign Exch. Reserves No Obs: Period: | 54% 2% 197 | 46% 98% 138 70Q1-2010Q4 |

Table 7: Variance Decomposition Table11 (Foreign Exchange Reserves, Reserve Money Shocks, and their Responses) cont'd...

 $^{^{11}}$ The reported variations are for the 10th period. The shocks are horizontal and the variables, which are affected from these shocks, are listed vertically.

4.2. The Oil Prices and M2 (reduced form OLS estimates)

This last section explores the effects of a change in oil prices on the money supply, where the results vary depending on the exchange rate regime.

The (reduced-form) OLS estimates of the restricted structural VECM suggest that there is a strong positive and statistically significant relationship between the oil prices and money supply for Oman, Nigeria, Venezuela, and Russia, with fixed/managed floating exchange rates. The positive response of reserve money from foreign exchange reserve shocks eventually make their way into the money supply for Oman, Russia, Nigeria, and Venezuela, which the reduced-form OLS estimates suggest.

Basically, oil exports affect the money supply of Oman, Nigeria, Venezuela, and Russia through bank reserves (see, Section 5.1.5), in common with Mundell (1963), Mussa (1976), Cuddington (1989), and Edwards (1986), who argue that foreign exchange reserve inflows from commodity booms will cause an expansion of the domestic monetary base and broader monetary aggregates (M2) in countries where there is either a fixed or a managed floating exchange rate. In Saudi Arabia, Kuwait, Qatar, Bahrain, and Iran, there is not a positive and a statistically significant relationship between the oil prices and M2, despite an increase in reserve money from foreign exchange reserves, which could be explained by some form of sterilization, in accord with Cuddington (1986), who argues that sterilization can reduce the money multiplier, and its larger effect on broader monetary aggregates.

In Canada and Norway, there is not a positive response of money supply to changes in oil prices, in common with Mundell (1963) and Mussa (1976), who argue that the money supply does not respond to an external shock, such as the oil prices, rather only does the exchange rate.

4.3. Order Dependence of Variables in the Structural VECMs

Are the VECM results order-dependent and robust?

A Structural VECM with Real Government Spending, Real GDP, Real Exchange Rates, and Real Oil (exogenous):

When the order of variable is changed from Real Government Spending, Real GDP, and Real Exchange Rates to Real GDP, Real Government Spending, and Real Exchange Rates, the results are insensitive to ordering for most of the oil exporters, except for Oman and Iran 12, for the reasons discussed below.

The new ordering of variables for Oman is counterintuitive to economic intuition, which shows that the government spending shock has a strong positive effect on output, in contrast to the more commonly observed and anticipated pro-cyclical fiscal policy response for the developing countries. (See, Kaminsky, Reinhart, and Vegh (2004)).

For Iran, the new ordering makes a clear difference to the policy outcome, with a positive effect of a real output shock on real government spending, which is more consistent with a pro-cyclical fiscal policy stance of developing countries, whereas before there was not a positive response of real government spending to a real output shock, with the initial order of variables.

The existing order of variables remains reasonable for a group of oil exporters. Under the Choleski decomposition, the order of variables goes from Real Government Spending to Real Output in the structural VECM equations of each country, where fiscal shocks are exogenous with respect to output, in common with Rotemberg and Woodford (1992), Fatas and Mihov (2001), Blanchard and Perotti (2002), Burnside et al. (2004) and Mountford and Uhlig (2008), who use the same order of variables in their VAR models.

A Structural VECM with GDP, M2, Inflation, and Oil (exogenous):

With a change in variables from GDP-M2-INF to M2-GDP-INF, the results are virtually the same, which allows me to conclude that the existing order of variables remains the most reasonable for all twelve oil exporters examined in this paper, in common with Bernanke and Blinder (1992), Bernanke and Mihov (1995), Christiano, Eichenbaum, and Evans (1994b), who assume the results are relatively insensitive to the ordering.

Next, I seek to find out whether a variable such as an economic activity is exogenous with respect to M2. For a variable to be exogenous, the variable cannot be affected either by contemporaneous or past values of any other endogenous variable, shown in the variance decomposition results. If one or more variables explain a significant portion of the forecast error variance decompositions of a variable at all forecast horizons, then the order of variables may not be correct, which may mean that GDP is not entirely exogenous with respect to M2. For UAE, Oman, Iran, and Nigeria, M2 shocks explain only a small amount of the fluctuations in GDP, which makes GDP exogenous.

Lastly, this paper tests for three key assumptions in the ordering of the endogenous variables, which has not been done before in the academic literature: a). policymakers have contemporaneous information about non-policy variables, such as GDP and CPI, whereas a policy variable, such as M2, is ordered last, with the equation ordered from GDP-CPI-M2, b). policymakers know only the lagged values of non-policy variables, with the order of variables that goes from M2-GDP-CPI, and c). policymakers have contemporaneous information about a non-policy variable

¹²The restricted VECM results with a new order of variables from real GDP to real government spending and real exchange rates are available upon request.

such as GDP, but a monetary aggregate, such as M2, predicts another non-policy variable CPI (implying that CPI is ordered last), with the order of variables that goes from GDP-M2-CPI, which is known as the existing order of variables.

Bernanke and Blinder (1992), Bernanke and Mihov (1995), Christiano, Eichenbaum, and Evans (1994b) use "assumption a" from above, where the non-policy macroeconomic variables, such as GDP and CPI, are ordered first before any other policy variable, such as a monetary aggregate, M2. The non-policy variables, GDP and CPI, affect M2 contemporaneously and through their lagged values, whereas only the lagged values of a policy variable, such as M2, affect the non-policy variables, such as GDP and CPI. The assumption here is that the policy shocks, such as the federal funds rate, total bank reserves, or the money supply do not affect any of the macroeconomic variables contemporaneously, but only with a lag. This identification scheme has been widely used in the developing countries.

I obtained relatively similar results under all three assumptions. However, the original ordering of variables (GDP-M2-INF) makes the most sense for a group of twelve major oil exporters.

5. Conclusion

The economies of oil-exporting countries may respond to oil shocks differently, depending on their political economic systems, including their regional economic alliance, stage of economic development, and the exchange rate regime. This paper tracks a vector of macroeconomic variables (real government spending, real output, real exchange rates, inflation, foreign exchange reserves, reserve money, and money supply), in response to oil price shocks, and analyzes whether any of these macroeconomic variables behave differently given the political economy factors. This paper concludes that the political economy factors explain why the economic indicators of the twelve major oil exporters to behave differently.

Most importantly, the focus of this paper is entirely on a group of oil exporters, unlike many other papers that focus on industrialized economies, which import oil. This paper separates oil exporters into two distinct categories, the GCC countries and the non-GCC oil exporters. This categorization of oil exporters is important, since the GCC countries share a political and economic alliance and their economies are largely oil dependent, with currencies (other than that of Kuwait) formally pegged to the US dollar. Unlike the GCC countries, the non-GCC oil exporters do not participate in any common political alliance or an economic union. Their economies are diversified and less dependent on oil exports. To my knowledge this type of categorization for oil exporters has not been done before in the academic literature.

I treat real oil as exogenous in the four-variable restricted structural VECMs of this paper, since no country influences the price of oil, maybe other than for Saudi Arabia. Such a treatment not reported in the academic literature results in a significantly high R2. I would like to add that other academic literature treats oil as endogenous 13.

This paper has reached several conclusions supported by economic reasoning and empirical evidence.

First, there is not a positive estimated response of real economic activity to real government spending shocks for eight out of the ten oil exporters (out of the ten countries with available government spending data), which suggests that countercyclical fiscal policy does not play a role in determining output for most the oil-exporting countries, except for Iran and Canada. These results are consistent with the economic reasoning that fiscal policies of developing countries tend to be pro-cyclical rather than countercyclical.

Some recommendation for future research would include whether there was any countercyclical fiscal policy implemented by developing oil-exporting countries for the 1970-2013 period, other than for Norway and Canada.

Second, real oil prices affect the real exchange rates of the GCC countries, but a real exchange rate appreciation for these countries with fixed exchange rate regime comes primarily in the form of inflation, rather than a nominal exchange rate appreciation. Contrary to the purpose of achieving low inflation under a fixed exchange rate regime, the GCC countries experienced significant inflation for the 1970-2013 period.

Third, money is endogenous in oil-exporting countries with fixed or managed floating exchange rates, including the GCC countries, Iran, and Venezuela; however, money is not endogenous in oil-exporting countries with floating exchange rates, such as Canada, Norway, and Nigeria.

Further research might extend the VAR analysis and empirical research to other countries with fixed or managed floating exchange rate regimes to find out whether money is endogenous for these countries and not for countries with floating exchange rates.

Fourth, there is a strong positive effect of oil price shocks on the foreign exchange reserves of the GCC countries, consistent with the restrictions of a fixed exchange rate regime under the conditions of a Mundell-Fleming model, whereas the effect of oil price shocks on foreign exchange reserves has been almost non-existent for the non-GCC oil exporters (except for Nigeria). This maybe because the non-GCC oil exporters (with flexible exchange rates) have significant leeway in investing their foreign exchange reserves.

Lastly, the foreign exchange reserves have a strong positive effect on the reserve money of the GCC countries, Venezuela, and Iran, which are countries with either fixed or managed floating exchange rates, whereas in countries with a purely floating exchange rate regime, such as that of Canada, Norway, and Nigeria14, the response of reserve

¹³See, Bernanke B., Gertler M., and Watson M. (1997), Blanchard and Gali (2007), Kilian (2009), and Kilian (2010) for use of endogenous oil prices.

¹⁴Nigeria adopted a floating exchange rate in 1986.

money to a foreign exchange shock has been almost non-existent, either due to weak or no impact of foreign exchange reserves on reserve money.

References

- Anshasy, A. (2014). "Oil shocks and oil producers' growth: where did all the spending go?" OPEC Energy Review.
- Altaee H. H. A. and Al-Jafari M. K. (2018). "Trade Openness and Economic Growth in the GCC Countries: A Panel Data Analysis Approach," International Journal of Business and Economic Sciences Applied Research, p. 62.

Bernanke B., Gertler M., and Watson M. (1997), 'Systematic Monetary Policy and the Effects of Oil Price Shocks', Brookings Papers on Economic Activity, 1, pp 91–142.

Bernanke, B.S. and A.S., Blinder, (1992). "The Federal Funds Rate and the Channels of Monetary Transmission," The American Economic Review, 82(4), pp. 901-21.

Bernanke, B. and I., Mihov (1998). "Measuring Monetary Policy," Quarterly Journal of Economics, 113(4).

Blanchard, O. and J. Galí (2007). "The Macroeconomic effects of oil price shocks: Why are the 2000s so Different from the 1970s? NBER Working Paper, No. 13368.

Blanchard, O. and Perotti (2002). "An Empirical Characterization of the Dynamic Effects of Changes in Government Spending and Taxes on Output." The Journal of Economics, November.

Burnside, C., M. Eichenbaum, and J. D. M. Fisher (2004). "Fiscal Shocks and Their Consequences." Journal of Economic Theory, 115(1): 89–117.

Christiano L., M. Eichenbau and C. Evans, (1994b). "Identification and the Effects of Monetary Policy Shocks", working paper WP-94-7, Federal Reserve Bank of Chicago (May).

Cuddington, J. (1989). "Commodity Export Booms in Developing Countries," World Bank Research Observer 4, 143-165.

Enders, W. (2008), 3rd Edition, "Applied Econometric Time Series," John Wiley & Sons, Inc.

Enders, Z., G. J. Müller, and A. Scholl. (2011). "How do fiscal and technology shocks affect real exchange rates?: New evidence for the United States." Journal of International Economics, 83, 53–69.

Engle, R. F. and C. W. J. Granger (1987), 'Co-integration and Error Correction: Representation, Estimation and Testing,' Econometrica 55(2), 251–76.

Fatás, A. and I. Mihov (2001). "Government Size and Automatic Stabilizers: International and International evidence," Journal of International Economics, Vol. 55, No. 1 3–28.

Frankel, J. (2010). "How can commodity producers make fiscal and monetary policy less pro-cyclical?" High Level Seminar on Natural Resources, Finance and Development, IMF Institute and Central Bank of Algeria.

Hamilton, J. D. (1994), Time Series Analysis, Princeton University Press.

Johansen, S. (1988). "Statistical Analysis of Co-Integration Vectors," Journal of Economic Dynamics and Control 12, 231-54.

Johansen, S. (1992). "Determination of Co-Integration Rank in the Presence of a Linear Trend," Oxford Bulletin of Economics and Statistics 54, 383–397.

Kaminsky, G., C., Reinhart, and C., Vegh (2004). "When It Rains It Pours: Pro-cyclical Capital Flows and Macroeconomic Policies," in NBER Macroeconomics Annual, ed. by Mark Gertler and Kenneth Rogoff.

Kilian, L. (2009). "Not All Oil Price Shocks are Alike: Disentangling Demand and Supply Shocks in Crude Oil Market?" American Economic Review, 99: (3), 1053-1069.

Kilian, L. (2010). "Oil Price Shocks, Monetary Policy and Stagflation," CEPR Discussion Paper.

Krugman, P.R. and Obstfeld, M. (2006) International Economics: Theory and Policy, 7th edition, Addison Wesley, Boston.

Lutkepohl, H. (2005). New Introduction to Multiple Time Series Analysis, Springer-Verlag, Berlin.

Marx, K. (1907). Capital. Vols. I, II, III. Chicago.

Minsky, H.P. (1990). "Endogeneity of Money," Hyman P. Minsky Archive, Paper 430.

Monacelli, T. and R. Perotti (2006). "Fiscal Policy, the Trade Balance, and the Real Exchange Rate: Implications for International Risk Sharing." Mimeo, IGIER.

Mountford A. and H. Uhlig (2008). "What are the Effects of Fiscal Policy Shocks?" NBER Working Paper No. 14551.

Mussa, M. L. (1976). "The exchange rate, the balance of payments, and monetary and fiscal policy

under a regime of controlled floating," Scandinavian Journal of Economics, 78, 229–248.

Mundell, R. A. (1963)."Capital mobility and stabilization policy under fixed and flexible exchange rates." Canadian Journal of Economic and Political Science 29(4): 4.

Nell, E.J. (2004). "Monetizing the Classical Equations: A Theory of Circulation," Cambridge Journal of Economics 2004, 28, 173-203.

Perotti, R. (2005). "Estimating the Effects of Fiscal Policy in OECD Countries." CEPR Discussion Paper No. 4842 (London: Centre , for Economic Policy Research).

Ravn, M. O., S. Schmitt-Grohé and M. Uribe (2012). "Consumption, Government Spending and the Real Exchange Rate." Journal of Monetary Economics, 59, 215–234.

Rotemberg, J. and M. Woodford (1992). "Oligopolistic Pricing and Effects of Aggregate Demand on Economic Activity. Journal of Political Economy. 1153–1207.

Sims, C.A. (1980). "Macroeconomics and Reality." Econometrica: Journal of the Econometric Society, 1-48.

Sims, C. A. (1986). "Are Forecasting Models Usable for Policy Analysis?" Federal Reserve Bank of Minneapolis,

Quarterly Review 10; 2-16.

Sims, C. A., J. H. Stock, and M. W. Watson (1990). Inference in Linear Time Series Models with Some Unit Roots. Econometrica 58:113-144.

This is an Open Access article distributed under the terms of the Creative Commons Attribution Licence

 \odot \odot